MICHIGAN ROSS SCHOOL OF BUSINESS

Working Paper

TARP Investments: Financials and Politics?

Ran Duchin Stephen M. Ross School of Business University of Michigan

Denis Sosyura Stephen M. Ross School of Business University of Michigan

Ross School of Business Working Paper Working Paper No. 1127 January 2010

This work cannot be used without the author's permission. This paper can be downloaded without charge from the Social Sciences Research Network Electronic Paper Collection: <u>http://ssrn.com/abstract=1426219</u>

UNIVERSITY OF MICHIGAN

TARP INVESTMENTS: FINANCIALS AND POLITICS*

Ran Duchin

Ross School of Business University of Michigan duchin@umich.edu Denis Sosyura

Ross School of Business University of Michigan dsosyura@umich.edu

First draft: June 26, 2009

This version: January 24, 2010

Abstract

We investigate the determinants of capital allocation to financial institutions under the Troubled Asset Relief Program (TARP). We find that banks' political ties play a significant role in TARP fund distribution. Connections to Congressmen on finance committees and representation at the Federal Reserve via board members are positively related to banks' likelihood of receiving TARP capital. The TARP investment amounts are positively related to banks' political contributions and lobbying expenditures. The effect of political influence is stronger for underperforming banks, thus shifting capital toward weaker institutions. Overall, we provide evidence about various channels through which political activism affects government spending.

^{*} We thank Kenneth Ahern, Oleg Chuprinin, Martjin Cremers, Ron Masulis, John Matsusaka, Raghu Rajan, David Robinson, and Stijn Van Nieuwerburgh for comments, as well as Satish Arcot and Gaurav Mani for research assistance.

Introduction

The Troubled Asset Relief Program (TARP), initiated in October 2008 to combat the financial crisis, authorized the U.S. Department of Treasury to invest federal capital in financial institutions. However, the legislation specified loose guidelines with respect to what assets will be purchased, how they will be valued, and how these funds can be used by financial institutions.¹

According to an independent valuation report by the Congressional Oversight Panel, the Treasury substantially overpaid for its investments under TARP in 2008, effectively subsidizing select banks.² In this paper, we study which institutions were more likely to receive these funds. In particular, we examine whether banks' political activism affected federal funding and whether capital investments were made in stronger or in weaker financial institutions.

Our main goal is to investigate the effect of banks' political ties, campaign contributions, and lobbying activity, if any, on the likelihood and amount of federal investments. One hypothesis is that banks with stronger political connections received favorable treatment in capital allocation. This view would be consistent with theories on the politics of government ownership and investment (for example, Shleifer and Vishny 1994), which suggest that federal capital would be used to accommodate private interests of politicians, such as transferring resources to favored institutions (for example, banks in politicians' voting districts). However, according to an alternative hypothesis, public scrutiny of banks' political actions, as well as the review and audit of TARP investments, would negate and even reverse the attempts to influence government decisions. For example, career concerns of federal officials under close monitoring

¹ The U.S. Government Accountability Office (GAO) provides the following summary about the use of TARP capital by largest participants: "With the exception of two institutions, institution officials noted that money is fungible and that they did not intend to track or report CPP capital separately." GAO Report to Congressional Committees, December 2008, p. 25.

² Valuing Treasury's Acquisitions: A Congressional Oversight Panel Report, February 6, 2009, p. 4-5.

(Fama 1980) represent one mechanism limiting the efficacy of banks' political activity. In fact, government officials may treat investments in connected institutions with extra caution to defend themselves against future accusations.

Our second goal is to investigate the relation between banks' financial strength and the likelihood of receiving TARP investments. One hypothesis is that capital investments were made in the strongest financial institutions with better earnings, lower risk, and higher capitalization. Retaining the healthiest banks in the financial system would be consistent with the theoretical framework in Atkinson and Stiglitz (1980) and Stiglitz (1993), in which government aid improves economic stability and social welfare. This view is also consistent with the program's objective to invest in banks that are "sound and not in need of government subsidization."³ On the other hand, anecdotal evidence from the financial media suggests that at least some federal investments were made in weaker banks to prevent their failure and avoid bank runs (Appelbaum 2008). According to these claims, banks with weaker earnings, lower capitalization, and higher risk would be more likely to receive federal funds. Under this scenario, government aid programs would subsidize underperforming institutions, resulting in misallocation of capital, an outcome predicted by Banerjee (1997) and Hart, Shleifer, and Vishny (1997). The government may also choose to subsidize larger (or more systemically important) institutions because their collapse would threaten the overall stability of the economic system.

Our empirical analysis focuses on the Capital Purchase Program (CPP), the largest TARP initiative by the number of participants and the amount of expended capital at the time of writing. As of September 22, 2009, 681 financial institutions received a total of \$204.6 billion under this program, an amount equal to 94 percent of the program's total projected investment funds.

³ Valuing Treasury's Acquisitions: A Congressional Oversight Panel Report, February 6, 2009, p. 5.

To measure political influence, we introduce four variables at the level of bank regulators and the federal government. Our first measure approximates political influence on the Federal Reserve, which is involved in the initial review of applications from the majority of qualified banks. We consider a qualified financial institution (QFI) to be connected to the Federal Reserve if an executive from this QFI holds a seat on the board of directors at one of the twelve Federal Reserve Banks or their branches.

Our second proxy is a bank's connection with a House member serving on the Congressional Committee on Financial Services and its subcommittees on Financial Institutions and Capital Markets, which played a significant role in the development of TARP and its amendments. We consider a bank connected with a House member if it is headquartered in the representative's election district. As another measure of political influence, we use a bank's sizeadjusted amount of campaign contributions to congressional candidates. Finally, our fourth proxy for political activism is size-adjusted lobbying expenditures by financial institutions.

We find that political connections are positively related to the likelihood of receiving CPP funds, controlling for other bank characteristics. This evidence appears whether political connections are measured via board seats at the Federal Reserve or ties with House members serving on key finance committees. The estimated magnitudes of political influence are substantial: a board seat at a Federal Reserve Bank is associated with a 31.3 percent increase in the likelihood of receiving CPP funds, while a bank's connection to a House member on key finance committees is associated with a 26.3 percent increase, controlling for other factors.

However, strong political connections might be indicative of banks' size and centrality (sample selection), or of their systemic importance. We are careful to consider these alternatives, and show that our results are robust to sample selection and banks' systemic importance. To address sample selection, we repeat our analysis after excluding the largest banks, banks

headquartered in New York, and banks with the best or worst financial condition, and obtain similar results. To accommodate the possibility that our political measures proxy for systemic importance, we reestimate our tests in samples matched on bank size and on bank size within the same home state, and obtain similar results. We also compute the systemic importance of financial institutions using Δ CoVaR, a measure proposed in Adrian and Brunnermeier (2009). We show that while systemically important institutions were more like to receive CPP funds, our conclusions remain unchanged.

Our findings also suggest that QFIs were more likely to receive an investment from CPP if they were bigger, had lower earnings, and had lower capital. This evidence is consistent with an investment strategy seeking to support systemically important institutions experiencing financial distress. We also find that the likelihood of receiving CPP capital is positively related to asset and management quality and negatively related to the amount of foreclosed real estate and mortgage-backed securities. One interpretation of these results is that these banks had a higher likelihood of surviving the crisis after receiving federal capital.

We also study the determinants of CPP investment amounts and find that the scaled amount of federal investment is positively related to banks' political contributions and lobbying expenditures scaled by bank size. A one standard deviation increase in political contributions to congressional candidates is associated with a \$14.6 million increase in allotted CPP funds. A one standard deviation increase in scaled lobbying amounts is associated with an additional \$10.4 million in raw CPP investment, controlling for other bank characteristics. Other political variables do not significantly explain investment amounts. One possible explanation is that bank regulators, such as the Federal Reserve, while plausibly helpful in supporting QFI applications, have little influence on the amount of funds awarded.

The amount of CPP investments is negatively related to capital adequacy, earnings, and liquidity, and positively related to bank size. This evidence is consistent with a strategy that seeks to strengthen the capital position of approved participants to a common acceptable level while providing an extra buffer for larger institutions that play a greater role in the stability of the financial system.

Interpreted broadly, these results show that political connections play an important role in a firm's access to capital. A natural question that arises is for which types of banks these connections play the greatest role. Our analysis indicates that the effects of political ties on federal capital investment are the strongest for companies with weaker fundamentals, lower liquidity, and poorer performance. This suggests that political ties shift capital allocation toward underperforming institutions.

Last, we also test whether the change in administration of January 2009 had an effect on the allocation of CPP funds. Our results do not reveal a reliable effect on the allocation of CPP funds. There are several possible explanations for this finding. First, the news about a change in administration preceded the allocation of the vast majority of CPP funds. Second, following the elections, the turnover in Congress was relatively small, and the changes in party affiliation of Congress members were even smaller. Third, the most politically active banks, as measured by political donations and lobbying expenditures, are typically connected to both Democratic and Republican parties.

The rest of this paper is organized as follows. Section 1 reviews related literature. Section 2 provides detail about the CPP, its investment criteria, and the application process. Section 3 describes our measures of political influence. Section 4 discusses the sample and data. Section 5 presents our main results. Section 6 discusses extensions. The conclusion provides summary and commentary.

1. Related Literature

To our knowledge, this paper is the first empirical investigation of the association between firm fundamentals, political influence, and the choice of TARP investments. At a broader level, it is part of a literature pioneered by Stigler (1971) and Peltzman (1976) that studies how politics influences economic regulation.⁴ More specifically, our research adds to the literature on the role of political factors in capital allocation, a topic that has been examined primarily in foreign markets.

For example, Khwaja and Mian (2005) show that politically connected firms in Pakistan borrow more from government banks (but not from private banks) and default more frequently on these loans. Sapienza (2004) finds that the interest rates charged by government-owned banks in Italy reflect the power of the party that controls the bank, resulting in preferential access to capital for party-affiliated borrowers. Faccio, Masulis, and McConnell (2006) provide evidence that politically connected firms are more likely to receive government funds in case of distress in thirty-five countries. Dinc (2005) finds that government-owned banks outside the United States increase their lending activity in election years to gain voter support. Our paper provides evidence on the relation between political ties and capital allocation in the United States during a financial crisis of historical magnitude. It considers a number of political channels simultaneously and compares their relative importance.

Our research also contributes to the literature on political activism and firm value and helps reconcile prior empirical findings. In previous studies, Roberts (1990), Fisman (2001), Faccio (2006), Cooper, Gulen, and Ovtchinnikov (2009), Faccio and Parsley (2009), and Goldman, Rocholl, and So (2009) find that corporate political ties have a positive effect on firm

⁴ A related line of empirical investigation is focused on the impact of corruption on aggregate outcomes such as growth and investment rates across countries (see Mauro 1995; Keefer and Knack 1995; Hall and Jones 1999; La Porta et al. 1999; and Glaeser and Saks 2006).

value. Others, such as Aggarwal, Meschke, and Wang (2009), argue that political activism is indicative of agency problems and show that firms with high political contributions experience significantly lower risk-adjusted returns. Our evidence suggests that political connections and activism serve as an insurance mechanism against extreme events. Thus, for example, the victory of a politician connected to the firm will increase the insurance value. However, this value will gradually decrease absent the occurrence of extreme events, explaining the underperformance of stocks of politically active firms compared to those of nonpolitically active companies.

Previous literature has also shown that politically connected firms have higher leverage ratios than their nonconnected peers (for example, Chiu and Joh 2004; Cull and Xu 2005; and Johnson and Mitton 2003). Our results are consistent with one possible explanation of these findings – namely, that lenders may rely on an implicit government guarantee that politically connected firms will be bailed out in a crisis.

We also add to the literature on the interaction between firms and congressional committees. Kroszner and Stratmann (1998) develop a positive theory of specialized, committee-based congressional organization that fosters ties between interest groups and legislators through reputation building. Our research supports this theory by showing that connections of financial institutions to specific congressional subcommittees affect the allocation of TARP funds.

Last, our research contributes to the growing literature on financial regulation during the credit crisis of 2008. Mian, Sufi, and Trebbi (2009) study politicians' voting behavior on key anti-crisis legislations in 2008 and show that representatives' voting patterns reflect the interests of their constituents and campaign contributors. We show that political influence extended well beyond the legislative process during the financial crisis and had a significant impact on the implementation of TARP. Veronesi and Zingales (2009) study the costs and benefits of TARP by examining the changes in enterprise value of the ten largest program participants around the

announcement of first capital infusions. They argue that the first TARP recipients realized substantial gains from the program and identify the banks that were the biggest winners. We find that, on average, later TARP recipients continued to benefit from TARP investments, as reflected in their positive announcement returns and that banks' political connections and financial performance affected their access to TARP funds. Next, several recent studies—Harvey (2008), Bebchuk (2009), and Coates and Scharfstein (2009)—critique the design of TARP and offer alternative ways to address the credit crisis. Although we do not discuss TARP alternatives, our results highlight one source of inefficiencies in the original TARP program. In particular, to the extent that political connections played a role in the distribution of TARP capital, this influence was the strongest for underperforming institutions, diverting funds toward weaker banks and constituting a departure from the program's declared objectives.

2. Emergency Economic Stabilization Act and the Capital Purchase Program

On October 3, 2008, the Emergency Economic Stabilization Act (EESA) was signed into law. The act authorized TARP – a system of federal initiatives aimed at stabilizing the U.S. financial system. According to the original plan, the Treasury intended to buy insolvent bank loans and mortgage-backed securities, hold these assets, and later sell them to private investors. However, on October 14, 2008, the government announced a revised plan—the Capital Purchase Program (CPP)—which authorized the Treasury to invest up to \$250 billion in financial institutions in exchange for their preferred stock and warrants. On March 30, 2009, the Treasury announced that it planned to expend \$218 billion of the \$250 billion authorized under this program.⁵

After the initiation of CPP, many other TARP initiatives were introduced. As of September 22, 2009, TARP included twelve financial programs. Yet CPP remains by far the largest by the amount of expended capital. This program is our primary focus.

⁵ SIGTARP Quarterly Report to Congress, April 21, 2009, p. 34.

Our focus on CPP is motivated by several reasons. First, CPP encompasses the greatest number of participating institutions among all TARP initiatives. Further, the program focuses on a relatively homogeneous group of firms—primarily domestically controlled banks, thrifts, and their holding companies. Next, whereas many TARP programs are still in their early stages, the vast majority of CPP funds has been invested. As of September 22, 2009, investments under CPP totaled \$204.6 billion, reaching 94 percent of the program's forecasted funding. Finally, whereas many TARP programs have evolved since their initiation, the rules and participation criteria for CPP have remained largely the same since the program's initiation in October 2008.

2.1 Qualifying Institutions and Their Primary Regulators

Financial institutions eligible to participate in CPP include domestically controlled banks, bank holding companies, savings associations, and savings and loan holding companies. For all banks and savings associations controlled by a holding company, CPP investments are made at the level of the holding company.

To apply for CPP funds, a qualifying financial institution (QFI) submits an application to its primary federal banking regulator. If the initial review is successful, the application is forwarded to the Treasury's investment committee, which makes a recommendation to the assistant secretary for financial stability, who makes the final decision regarding the investment. CPP is administered by the Treasury's Office of Financial Stability, which was created by the EESA.

Primary federal banking regulators include the Federal Reserve, the Federal Deposit Insurance Corporation (FDIC), the Office of the Comptroller of the Currency (OCC), and the Office of Thrift Supervision (OTS). Each of the four regulators has oversight responsibilities over different types of banking institutions, which are discussed in more detail in Appendix I.

2.2 Application Process and Selection Criteria

The application process begins with the submission of a two-page CPP application form to the QFI's primary regulator. In cases where the QFI is a bank holding company, the application is submitted both to the regulator overseeing the largest bank of the holding company and to the Federal Reserve. This detail is important for our study, since it provides the Federal Reserve an opportunity to review applications from the vast majority of QFIs. Before submitting an application, banks are encouraged to consult with their banking regulator whether to apply.⁶

After receiving an application, the banking regulator evaluates the financial strength and long-term viability of the QFI based on a set of standardized ratios and examination ratings. Although the specific weighting of factors is not made public, bank regulators generally use the Camels rating system to assess an institution's financial condition. The system derives its name from the six factors that are evaluated: *Capital adequacy*, *Asset quality*, *Management*, *Earnings*, *Liquidity*, and *Sensitivity* to market risk. The ratings are based on financial statements and onsite examinations by regulatory authorities. A score from 1 (best) to 5 (worst) is assigned for each of the six factors, and these scores are aggregated to derive an overall score of a bank's financial health. The weighting of factors and banks' combined ratings are not made public, even on a historical basis.

Applications from banks with the strongest ratings receive presumptive approval from the banking regulators and are forwarded to the Investment Committee at the Treasury's Office of Financial Stability (OFS).⁷ Applications from banks with mediocre ratings are referred for further review to the CPP Council, which considers other factors that may outweigh the lower ratings, such as private equity investments in the institution or a pending merger of the applying

⁶ "Troubled Asset Relief Program: Additional Actions Needed to Better Ensure Integrity, Accountability, and Transparency," GAO Report to Congressional Committees, December 2, 2008, p. 23.

⁷ The Investment Committee consists of five investment officers: OFS's chief investment officer (committee chair) and the assistant secretaries for financial markets, economic policy, financial institutions, and financial stability.

institution.⁸ Last, QFIs with low ratings are encouraged to withdraw their applications. All application submissions and withdrawals are kept confidential to protect banks' interests.

In the final stage of the process, OFS's Investment Committee reviews applications forwarded by the banking regulators and the CPP Council. The committee makes a recommendation to the assistant secretary for financial stability, who makes the decision about the investment. If the approval of the application is unlikely at this stage, the application can still be withdrawn.

2.3 Financial Conditions of the Program

In exchange for CPP capital, banks provide the Treasury with cumulative perpetual preferred stock. The preferred shares are issued at par and are equivalent in seniority to the most senior type of the QFI's preferred stock. The shares pay quarterly dividends at an annual yield of 5 percent for the first five years and 9 percent thereafter. The amount of the investment in preferred shares is determined by the Treasury, although QFIs state the requested amount in the CPP application.

In addition to preferred stock, the Treasury obtains warrants for the common stock of publicly traded QFIs. The warrants, valid for ten years, are issued for such number of common shares that the aggregate market value of the covered common shares is equal to 15 percent of the investment in the senior preferred stock. The initial exercise price for warrants is determined as the average closing price of common shares over the trailing twenty trading days ending on the date of the investment in preferred shares. As of September 22, 2009, no common stock warrants in public companies have been exercised.

⁸ The CPP Council consists of representatives from the four federal banking regulators and Treasury officials, who act as observers.

In cases of private companies, the Treasury receives warrants to purchase additional shares of preferred stock that pay dividends at a higher annual rate of 9 percent. These warrants have an exercise price of \$0.01 and are exercised immediately to entitle the Treasury to dividends on the new shares. Appendix II offers additional detail on the terms of capital redemption, conditions imposed on participating institutions, and investments in "S" corporations and mutual organizations.

2.4 Oversight

The execution of TARP, including CPP investments, is overseen by several monitors: the Financial Stability Oversight Board (FSOB), the Congressional Oversight Panel, the Office of the Special Inspector General (SIGTARP), and the Comptroller General. Appendix III provides the details about the authority and structure of these entities.

Despite the overseeing authorities, many officials have acknowledged a lack of transparency in the Treasury's investment decisions. To illustrate, in his confirmation hearing at the Senate in January 2009, Treasury secretary nominee Timothy Geithner admitted: "There are serious concerns about transparency and accountability, confusion about the goals of the program, and a deep skepticism about whether we are using the taxpayers' money wisely."⁹ A similar issue is voiced in SIGTARP's quarterly report to Congress released on April 21, 2009, which describes the initiation of an audit to investigate the external influence on the decisions of the Treasury and primary bank regulators. In the next section, we introduce several variables that seek to measure political influence of interested parties on the Treasury and bank regulators.

⁹ Confirmation Hearing Statement of Treasury secretary nominee Timothy Geithner before the U.S. Senate Committee on Finance, January 21, 2009.

3. Measures of Political Influence

We construct several measures of political influence. At the level of bank regulators, we use board memberships at the Federal Reserve as a proxy for the potential influence on the decisions of a banking regulator. At the level of the federal government, we use three measures. The first is QFIs' representation on key congressional committees involved in oversight of banking and capital markets. The second is QFIs' political contributions to congressional candidates in the 2008 election cycle, scaled by bank size. The third is lobbying expenditures in 2008, also adjusted for bank size. This section describes each type of connection.

3.1 Federal Reserve Boards of Directors

Among the four banking regulators that conduct the initial review of CPP applications, we focus on the Federal Reserve for two reasons. First, this regulator is responsible for the oversight of bank holding companies and state-charted member banks, which comprise over 85 percent of QFIs that received CPP investments. Second, unlike other regulators, boards of directors at Federal Reserve Banks include QFI executives, providing them with significant influence and authority. For example, board members at a Federal Reserve Bank appoint presidents and vice presidents of their bank, as well as all its officers, and oversee the bank's internal audit program.¹⁰

QFI executives serving on the boards of the main offices of Federal Reserve Banks are elected by member banks in each district for three-year terms. In the election, banks are divided into three categories by the amount of capital—large, medium, and small—and each category elects one director from member institutions. This detail is important because it implies that our

¹⁰ The appointment of Federal Reserve Banks' presidents and vice presidents is subject to approval by the Board of Governors.

measure of connections to the Federal Reserve board members is not biased toward larger banks. Appendix IV provides additional detail on Federal Reserve directors, their tenure, and affiliation.

We consider a QFI connected to the Federal Reserve if an executive or an officer of this QFI served on the board of directors of a Federal Reserve Bank or its branch in 2008 or 2009. In our sample, 83 financial institutions (1.1 percent of our sample) have such a connection. The data on board composition are collected from annual reports of Federal Reserve Banks and the Board of Governors.

3.2 Bank Representation on the House Financial Services Committee

At the level of Congress, we investigate the ties between QFIs and the members of the House Financial Services Committee and its two subcommittees: the Subcommittee on Financial Institutions and the Subcommittee on Capital Markets.¹¹ The Committee on Financial Services reviews and develops legislation in all areas of the financial system. This committee played a key role in the development of the EESA and continues to exert strong influence on amendments related to expanding and modifying TARP, which require the approval of Congress.

Within the Financial Services Committee, we focus on two subcommittees, which oversee banking and financial markets. The Subcommittee on Financial Institutions supervises all primary banking regulators and matters related to the soundness of the banking system. The Subcommittee on Capital Markets reviews laws regulating investment banks and capital markets, among other issues. The House Financial Services Committee comprises four more subcommittees, whose jurisdiction covers other finance issues, such as monetary policy, international trade, and technology.

¹¹ The full names of the two subcommittees are: Subcommittee on Financial Institutions and Consumer Credit, and Subcommittee on Capital Markets, Insurance, and Government Sponsored Enterprises.

We focus on the committees in the House of Representatives rather than on those in the Senate for several reasons. First, election districts for the House are substantially smaller, resulting in a tighter link between a politician and QFIs headquartered in his or her district.¹² Second, districts of all representatives are approximately equal by the number of voters (about 700,000 voters per district), providing an equitable comparison base in the cross-section. Last, our focus on the House of Representatives enables us to cover all eligible QFIs, including those located in Washington, DC, Puerto Rico, and U.S. territories, which elect delegates to the House but have no representation in the Senate.

To identify House members representing the district of each QFI, we use the zip code of the institution's headquarters. We collect data on congressional representatives who held office in 2008 (110th Congress) and in 2009 (111th Congress). Overall, approximately 87 percent of House representatives from the 110th Congress were reelected in 2008 for the next term. Our combined sample for 2008 and 2009 includes 499 members.

Using congressional records, we identify the members of the House Financial Services Committee and its subcommittees on Financial Institutions and Capital Markets. Next, for each QFI we calculate a measure of representation on key committees. This proxy is computed as the average of six indicators, each equal to one if the representative from the QFI's district served on the Financial Services Committee, the Subcommittee on Financial Institutions, or the Subcommittee on Capital Markets, respectively, in either 2008 or 2009. Thus, we have three indicators for each of the two years. For example, if a representative served on the Financial Services Committee and the Financial Institutions Subcommittee but not the Capital Markets

¹² Unlike senators, who are elected in a statewide election, House representatives are elected in smaller districts within each state, with an average of nine election districts per state.

Subcommittee in both 2008 and 2009, a QFI from this representative's district will be assigned a score of 0.67.¹³ In the empirical section, we disentangle the effects of each committee.

The Financial Services Committee includes 71 members, and its subcommittees on Financial Institutions and Capital Markets comprise 45 and 50 members, respectively. A total of 1,205 QFIs (16.6 percent of our sample) have a representative who is a member of at least one of the three groups.

To illustrate the power of the Committee on Financial Services, consider the following example, published in the *Wall Street Journal* on January 22, 2009 (Paletta and Enrich 2009). In late September 2008, Boston-based OneUnited Bank found its capital depleted and its management compromised with a cease-and-desist order from the FDIC for poor lending practices and the abuse of management pay. Yet in mid-December 2008, the bank received \$12,063,000 from CPP.

The bank is based in the home state of Rep. Barney Frank, head of the Financial Services Committee, who acknowledges that he had included into TARP a provision aimed at helping this particular bank and recommended to regulators that OneUnited be considered for capital investment under TARP. The bank's lawyer admitted that he had discussed the bank's financial situation over the phone not only with Frank but also with Rep. Maxine Waters (a member of the Financial Services Committee and Financial Institutions Subcommittee), whose husband used to be a OneUnited director.

3.3 Political Campaign Contributions

In addition to electoral ties that exist between politicians and banks headquartered in their election districts, companies can establish connections with politicians by financing their election campaigns. The financial sector is one of the most active contributors to federal political

 $^{^{13}}$ 0.67 = (1+1+1+1+0+0)/6

campaigns. In 2008, companies in the financial sector contributed over \$475 million to congressional candidates, the largest amount ever contributed by this industry and the second-largest figure among all U.S. industries.¹⁴ To study the influence of political donations, we collect data on contributions made by financial institutions to Political Action Committees (PACs), as well as individual donations initiated by banks' executives, employees, and their spouses in the 2008 election—the most recent election cycle preceding TARP.

Political Action Committees are private groups organized to support political candidates. Corporations are prohibited from making direct campaign contributions but can sponsor PACs by providing them with financial resources. PACs, in turn, invest these resources by making donations to political candidates and committees, consistent with the interests of the PAC's constituents. For example, in the 2008 election cycle, the Wells Fargo Employee PAC received \$1.3 million in contributions, primarily from Wells Fargo executives and managers. From these funds, the PAC donated \$767,000 to support federal political candidates.

Individual donations to political campaigns come directly from corporate employees and their spouses. These contributions are currently limited to \$115,000 over a two-year cycle, of which a maximum of \$45,600 can be contributed to all federal candidates, and a total of \$69,900—to all PACs and parties. Also, the Federal Election Commission (FEC) establishes a cap of \$2,400 on contributions to any one candidate in a given election and a maximum of \$5,000 per calendar year to any one PAC.

We include individual donations of banks' employees and their spouses as part of total political contributions made by the financial institution. We establish this connection based on the reported company affiliation of the contributing individual. Individual contributions account for the majority of donations to federal political candidates.

¹⁴ Center for Responsible Politics, 2009.

All types of political donations exceeding \$200 must be reported to the FEC, which collects these data and makes them available to the public. Our data come from the Center for Responsive Politics (CRP), a nonprofit organization that aggregates FEC filings and provides a searchable database of contributions by company and politician. In cases when companies merge during our sample period, we attribute the total spending by the merging companies to the combined firm. Our data enable us to track donations to particular politicians, providing an opportunity to study the influence of the recipient's seniority, committee membership, and other details. We offer this analysis in the empirical section.

Our sample of political contributions includes 1,668 qualified financial institutions. As shown in Table I, an average (median) QFI donated a total of \$36,228 (\$2,000) to congressional candidates in the 2008 election cycle. In our empirical analysis, we scale raw amounts of contributions by bank assets to adjust for cross-sectional differences in size.

3.4 Lobbying Expenditures

Lobbying refers to the practice of petitioning government agencies. As a form of political influence, lobbying has several distinct features. First, it is usually focused on government legislation or specific issues rather than on particular politicians. Second, most companies lobby their interests by hiring an external lobbying firm, which works with government officials on behalf of its clients.

According to the Lobbying Disclosure Act of 1995, organizations with annual lobbying expenditures greater than \$25,000 must register with the Senate's Office of Public Records and submit lobbying reports, which are available to the public. Since lobbying reports are filed by lobbyists rather than their clients, the threshold of \$25,000 in annual expenditures obligates virtually all lobbying firms to file such reports, even if the annual spending of a particular client is below the threshold.

We collect lobbying data from the CRP, which maintains a database of lobbying reports. Each report provides information on the issues lobbied, the government agencies petitioned, and the amount spent.¹⁵ The issue and agency classifications include 78 and 311 categories, respectively.

In our base specification, we use the amount of bank's lobbying expenditures in 2008 through the first quarter of 2009, scaled by bank assets. In cases when lobbying is done by a bank held by a holding company or by a bank subsequently acquired before September 30, 2008, the total lobbying amount is attributed to the combined firm. Our sample includes 2,204 reports submitted on behalf of 76 QFIs. As shown in Panel A of Table I, the average (median) amount of total lobbying expenditures in our sample in 2008–2009 is \$5.08 million (\$1.09 million) per QFI. In more detailed analysis, we use additional filters, which cover only the reports in which the category of issues lobbied includes banking or finance, and the agencies lobbied include the Treasury or one of the four primary banking regulators.

4. Sample and Data

4.1 Qualifying Financial Institutions

To construct a sample of QFIs, we begin with a list of all FDIC-insured financial institutions that were active as of September 30, 2008. This group includes 8,391 institutions. Because CPP capital is invested at the level of holding companies, we treat institutions held by the same holding company as one QFI and aggregate their financials.¹⁶ The vast majority of holding companies (91 percent) own only one institution. Independent institutions, which are not held by

¹⁵ Lobbyists are obligated to disclose the category of the issues lobbied if the amount of lobbying expenditures exceeds \$5,000.

¹⁶ We ascertain the correctness of matching between institutions and holding companies by cross-referencing the FDIC records with the data from the Federal Reserve, both of which provide the ultimate parent for banking institutions.

higher holders, are treated as individual QFIs. Overall, there are 7,343 holding companies and independent institutions in our sample.

Following the CPP eligibility criteria, we exclude 49 banks or holding companies with foreign control.¹⁷ We also eliminate institutions that made a public announcement that they would not apply to CPP. To identify these institutions, we search the Press Release News Wire and news publications in the Factiva database for different variations of an institution's statement that it would not apply to CPP or that it would decline CPP funds. We can identify 94 institutions that made such statements, of which 39 announced their decision not to apply and 55 applied to CPP and were approved but decided to decline the investment. We exclude the 39 institutions that declined to apply. For the purposes of our tests, we treat the 55 declining institutions that were approved as CPP recipients. As a result of these filters, we end up with 7,255 QFIs in our sample, whose summary statistics are provided in Panel A of Table I.

The data on the recipients of CPP funds are collected from the Office of Financial Stability, which provides the amount of funds invested, the date of investment, and the consideration received by the Treasury in exchange for capital. As of September 22, 2009, 681 financial institutions received a total of \$204.6 billion under this program. The distribution of CPP investment amounts is heavily right-skewed, and the average (median) CPP investment (including QFIs that were approved and later declined CPP funds) is \$301.3 (\$11.4) million, as shown in Panel A of Table I.

One challenge in our sample construction is that we cannot identify the specific institutions that applied to CPP, since these data are kept confidential. According to the GAO

¹⁷ For the purposes of CPP, control is defined as direct or indirect ownership of at least 25 percent of a bank's shares.

report to Congress, thousands of CPP applications were submitted.¹⁸ This estimate provides some comfort that our sample, though undoubtedly larger, is of similar order of magnitude to the pool of CPP applicants.

Aside from the sample size, another challenge arises from possible self-selection of CPP applicants. In particular, some banks may have chosen not to apply for CPP funds, for example, because they did not need the extra capital. To this extent, we eliminate the banks that publicly announced their decision not to apply to CPP, but cannot ascertain that this list is comprehensive. However, it should be noted that the concern about sample selection is mitigated by the attractive conditions of CPP and the below-market cost of CPP capital, which provided a strong incentive even for well-capitalized institutions to participate in the program. For example, Veronesi and Zingales (2009) show that CPP funds increased enterprise value of the ten largest recipients by \$131 billion, effectively providing a value transfer from taxpayers to these banks. A qualitatively similar conclusion was reached by an independent valuation report of the Congressional Oversight Panel, which determined that the Treasury paid a premium of 34 percent on its largest CPP investments.¹⁹

Consistent with the attractiveness of participation in CPP, Panel B of Table I shows positive announcement returns for publicly traded bank holding companies around the dates of their approval for CPP capital. Across all specifications and announcement windows, stock price reaction to this news is virtually always reliably positive.²⁰ Return magnitudes are also

¹⁸ "Troubled Asset Relief Program: Status of Efforts to Address Transparency and Accountability Issues." GAO Report to Congressional Committees, January 30, 2009, p.13.

¹⁹ Valuing Treasury's Acquisitions: A Congressional Oversight Panel Report, February 6, 2009, p. 4–5.
²⁰ To test whether the value creation from CPP capital infusion can be generalized to healthier banks, we also estimated market returns around the announcement of banks' approval for CPP funds for a subsample of well-capitalized banks, as per bank regulators' definition based on the capital adequacy ratio. Our results (unreported) indicate that announcement of CPP approval is associated with positive returns even for well-capitalized institutions. We also repeated this analysis for other measures of financial condition, such as profitability, and obtained similar results. Overall, the above evidence suggests that the below-market cost of CPP capital made it attractive even for healthy institutions, thus somewhat alleviating the concern about a potential bias in our sample.

nontrivial: around 1 percent for a one-day window [-1; 0] and between 1.3 percent and 2.5 percent for a three-day window [-1; 1].

Last, in the robustness section we conduct additional tests to address the issue that the decision to apply for CPP capital is unobservable. We repeat our analysis for various subsamples of financial institutions to mitigate the concern that our results may be driven by sample selection. We also test the robustness of our results to various measures of banks' systemic importance. In particular, we replicate our tests using size-matched subsamples of politically connected and nonconnected institutions. We also control for systemic risk of financial institutions based on a measure developed in Adrian and Brunnermeier (2009). These tests are discussed in the robustness section.

To provide another perspective on the distribution of CPP funds, Figure I illustrates the number of CPP recipients and the total amount of investment across different U.S. regions. Panels A and B show the raw number of CPP recipients and the fraction of qualifying institutions headquartered in each state that received CPP capital, respectively. Panels C and D provide evidence on the amounts of CPP investment, with Panel C showing raw amounts and Panel D presenting amounts scaled by the size of qualifying institutions. The scaled figures of CPP recipients and amounts (Panels B and D) provide a similar pattern. Overall, the distribution of CPP funds is skewed toward the states with greater population, such as California and the densely populated states in the Northeast and Mid-Atlantic, even after adjusting for the greater number of banks headquartered in more populated regions.

From a political angle, the evidence presented above is consistent with a scenario in which more populated states benefit from wider representation in Congress and greater importance for attracting electoral votes. Table II provides more formal descriptive evidence in

this direction by demonstrating a strong positive relation between state-level representation in Congress and CPP capital allocation.

Of course, it is possible that more populated states provide homes to banks that meet the criteria for CPP or are simply more systemically important. To study the relation between bank fundamentals and the distribution of federal aid, we construct an array of financial measures of banks' condition and performance. These financial data are collected from the reports of condition and income filed by all active FDIC-insured institutions every quarter.

4.2 Financial and Demographic Variables

Financial Variables

In constructing our financial variables, we seek to follow the Camels assessment criteria used by bank regulators. Since the data on the actual ratings assigned by bank regulators are never made public, we introduce proxy variables for each of the six categories in the Camels system: *C*apital adequacy, *A*sset quality, *M*anagement, *E*arnings, *L*iquidity, and *S*ensitivity to market risk. Our choice of proxies is generally guided by the financial indicators employed by the FDIC in Uniform Bank Performance Reports that evaluate banks on similar dimensions. Condition ratios are computed as of September 30, 2008, the latest quarterly report before the CPP application deadlines for public and private institutions. Detailed descriptions of the financial variables are summarized in Appendix V.

Financial Strength Index

In addition to analyzing each dimension of the Camels system, we compute a bank's aggregate strength index. For each bank, we calculate its ranking in our sample on each Camels proxy and find the mean of these rankings to derive an aggregate index score, normalized to lie between 0 and 1. For example, a strength index of 0.90 indicates that a bank's average ranking on the

Camels proxies places it in the ninetieth percentile of our sample—that is, the top 10 percent of QFIs.

Demographics

Our demographic variables include bank size and age. Size is measured by the natural logarithm of assets. As Panel A of Table I shows, the average (median) QFI in our sample holds assets worth \$1.7 billion (\$144 million) and is 71 (81) years old. We expect that larger banks are more likely to receive CPP investments due to their greater role in the financial system. Bank age is included to test the hypothesis that more established banks are more likely to be saved.

Last, Panel C of Table I provides the correlation matrix for our independent variables. The vast majority of correlations are low, indicating that our measures capture distinct political, financial, and demographic characteristics. In the next section, we study the relation between these variables and the likelihood of approval for CPP funds.

5. Main Results

This section presents our main empirical findings. We begin with nonparametric evidence and continue with regression analysis of the effect of political, financial, and demographic factors on the likelihood of CPP investment. The section concludes with a discussion of alternative hypotheses and tests of robustness.

5.1 Univariate Evidence

Table III provides nonparametric comparison of the likelihood of receiving CPP capital between QFIs evaluated on three sets of indicators: political connections, financials, and demographics. For each measure of political connections, we distinguish between connected and nonconnected QFIs. For each financial and demographic variable, we compare banks above and below the median value.

Table III shows that institutions with political connections were more likely to receive CPP capital. This finding holds across all four measures of political ties, and the differences between connected and nonconnected QFIs are significant at the 1 percent level.

Institutions with weaker financial indicators were also more likely to receive CPP investments. This finding holds for all Camels proxies, and the differences between "strong" and "weak" companies are statistically significant at the 1 percent level, except for sensitivity to market risk. The evidence on the strength index also indicates that QFIs with lower index values were more likely to receive CPP funds. Finally, bigger banks were also more likely to receive CPP funds.

Nonparametric evidence so far suggests that stronger political connections and weaker financial indicators are associated with a higher likelihood of CPP investment. However, the univariate tests consider variables in isolation and do not account for their interaction. In the next section, we provide regression evidence in order to examine the independent variables together and to compare their relative importance.

5.2 Multivariate Analysis

We estimate cross-sectional logit regressions to determine the influence of bank characteristics on the likelihood of receiving CPP capital. The dependent variable is an indicator equal to 1 if a financial institution was approved for CPP funds and 0 otherwise. The independent variables include four measures of political connections, and institutions' financial and demographic indicators, described above. To control for the possibility of systematic differences in decision criteria across bank regulators, we include regulator fixed effects. Standard errors are clustered at the company level.

The results of estimating the likelihood of CPP investment are summarized in Table IV. The first four columns consider each of the political variables individually, while controlling for

company-level financials and demographics. Column (5) presents the results of the joint effect of all political variables. Column (6) shows the effects of including all political, financial, and demographic variables simultaneously.

Across these different specifications, representation on congressional finance committees and board membership at Federal Reserve Banks are strongly positively related to the likelihood of receiving CPP investments, and these relations are significant at the 1 percent level. The effect of campaign contributions, though positive, is noticeably weaker. It is possible that the scrutiny of contribution records available to the public motivates financial institutions to rely more heavily on other channels of political influence. The effect of lobbying, though also positive, is statistically insignificant after controlling for bank fundamentals. This result may be explained by the fact that we cannot identify the cases of lobbying specifically for TARP funds, given a relatively general classification of issues in lobbying reports. It is also possible that we are unable to reliably detect the marginal effect of lobbying due to the small fraction of lobbying firms in our sample.

The influence of political variables is also economically significant. Based on the estimates in column (6), a board seat held by a QFI executive at a Federal Reserve Bank increases the probability of CPP investment by 31.3 percent for a bank with mean fundamentals. A mean bank headquartered in the district of a representative serving on key finance committees in Congress was 26.3 percent more likely to receive CPP funds, controlling for fundamentals.

The evidence from financial variables indicates that QFIs were more likely to receive CPP capital if they were bigger, had lower earnings, and had lower capital. These findings are consistent with an investment strategy seeking to support systemically important institutions experiencing financial distress.

We also find that the likelihood of receiving CPP capital is positively related to asset quality and negatively related to the proportion of foreclosed real estate. A combination of lower trailing earnings and weaker capital position, along with higher asset quality, could be a result of writing off or selling troubled assets. Under this interpretation, CPP investments were made in the banks that had reduced their forward-looking exposure to the crisis and were likely to survive after receiving federal capital. QFIs with better management were also more likely to receive federal funding. This is consistent with investing in institutions that are likely to make better use of the money in order to withstand the crisis.

Column (7) of Table IV evaluates the overall effect of a bank's financial strength on the likelihood of CPP investment. In this specification, the financial strength is captured by the financial strength index, measured as the bank's average rank on the six Camels proxies. This analysis seeks to distinguish between two hypotheses regarding CPP investment objectives: (1) strengthening the financial system by investing in the healthiest institutions; and (2) bailing out distressed banks to prevent their failure and avoid bank runs.

As shown in column (7) of Table IV, the coefficient on the strength index is negative and strongly significant with a t-statistic of 7.1. This evidence is more consistent with the latter hypothesis, although it should be interpreted with caution, given the nonuniform direction for individual measures of financial condition. The results on the strength index are also consistent with the industry evidence from banks participating in CPP.²¹

²¹ For example, according to Joseph DePaolo, the CEO of Signature Bank, which received \$120 million from CPP in December 2008: "It's Congress' fault for allowing it to go from a program for healthy institutions to a program that they're giving money to weak institutions." Interview with National Public Radio, April 1, 2009, available at http://www.npr.org/templates/story/story.php?storyId=102618967

5.3 Robustness

In this section, we test the robustness of our findings to alterative explanations. We start by considering different measures of bank fundamentals and then move to the critical issues of sample selection and systemic importance in the financial system.

The first issue we consider is alternative measures of bank fundamentals. So far our choice of financial variables has been guided by the Camels system developed by banking regulators. Yet each dimension of the Camels system can be measured by a different set of proxies. Therefore, we would like to test the robustness of our results to a different choice of condition and performance indicators. These alternative measures are discussed in Appendix V. Columns (1) and (2) of Panel A in Table V present the estimation results with new sets of financial variables. The qualitative conclusions for all political variables and the majority of financial variables remain unchanged, suggesting that our main results are consistent across different measures of fundamentals.

The next alternative explanation we consider is sample selection. Our sample comprises all banks eligible for CPP investments, not just banks that applied for CPP funds. It is possible that the institutions that applied to the program were in distress and had the greatest need for capital. Under this scenario, the true sample of CPP applicants represents a subsample of QFIs with substantially weaker fundamentals. This alternative hypothesis would explain the negative relation between the likelihood of CPP investment and capital adequacy, earnings, and the overall indicator of banks' financial health.

To test this hypothesis, we eliminate the top 25 percent of institutions based on capital adequacy (tier-1 capital ratio) and estimate logit regressions with all political, financial, and demographic variables using the remaining QFIs. The results of this estimation are presented in

column (3) of Panel A in Table V. Overall, the signs, magnitudes, and significance levels of all coefficients remain very similar to the main specification.

As another check, we eliminate the top 25 percent of QFIs based on the financial strength index and repeat the estimation. The results, presented in column (4) of Panel A in Table V, are also qualitatively similar to our main specification (column [6] in Table IV). Altogether, this evidence suggests that our findings continue to hold after excluding the least likely CPP applicants.

Next, we consider the critical issue of systemic importance. Since the primary declared objective of TARP was achieving financial stability, it is possible that our results are explained by banks' systemic importance. In particular, the first potential concern is that our results may be driven by the well-connected largest QFIs, which can be deemed too big to fail, even if they have weaker fundamentals, lower capital ratios, and poor financial performance. To test whether our results are driven by the largest institutions, we exclude the top 1 percent of banks (73 QFIs), which account for 78.3 percent of all QFI assets. Among the largest banks, 68.5 percent received CPP funds, consistent with the "too big to fail" argument. Column (5) of Panel A in Table V shows the results of estimating our logit model after excluding the largest banks. All qualitative conclusions hold, and the quantitative results are also similar.

As an additional robustness test, we also exclude all 173 QFIs headquartered in New York City. As America's financial center, New York might be more likely to elect representatives with prior finance experience, who, arguably, are also more likely to serve on the House Financial Services Committee because of this experience. Thus, even if New York banks receive capital because they are more critical to the stability of the financial system, our tests might imply that they are more likely to receive CPP funds because of greater representation on the finance committees in Congress.

To test this alternative explanation, we repeat our estimation after excluding all QFIs headquartered in New York City. This exclusion has little effect on our results, as shown in column (6) of Panel A in Table V. The variables capturing Congress finance committee representation, which are determined by location, remain strongly positively related to the likelihood of CPP investment. The signs and significance of financial variables also remain largely unchanged.

It is also possible that the effect of systemic importance extends well beyond the largest, centrally located banks and plays a significant role in CPP investment in midsize financial institutions. To distill the effect of political connections from that of systemic importance, we construct sized-matched subsamples of politically connected and unconnected banks. For each measure of political activism, and for each connected bank based on that measure, we identify an unconnected bank closest in size (total assets). To ensure equitable size within each pair, we use only observations for which the difference in size between a connected firm and its unconnected match does not exceed 10 percent. This procedure generates four subsamples of connected and unconnected firms, with each sample corresponding to one measure of connections. We also generate a combined size-matched subsample for firms that are connected based on at least one measure of connections. The bottom row in Panel A of Table VI provides t-statistics for the difference in mean size between connected firms. As expected, the t-statistics are close to zero across all matched groups.

Next, we reestimate our base logit specification in each of the five size-matched subsamples of connected and unconnected firms. The effect of political measures remains statistically significant and qualitatively similar despite the noticeable reduction in sample size as a result of the relatively restrictive size matching. In particular, board representation at the Federal Reserve and connections to Congress representatives on finance committees are

positively related to the likelihood of CPP investment, and this relation is significant at the 5 percent level or better. The two other measures of connections—lobbying and political contributions—remain positive but insignificant at conventional levels, as reported in the main analysis. Finally, consistent with our main findings, possession of at least one type of political connection has a positive effect on the likelihood of CPP investment, a result significant at the 1 percent level (column [5] of Panel A in Table VI). Overall, to the extent that asset size captures systemic importance of a financial institution, our results hold after controlling for systemic importance via both parametric and nonparametric methods.

It is also possible that asset size alone is an insufficient statistic to capture systemic importance. For example, as a result of variation in the number of banks across different states, a smaller financial institution may be more important for its regional financial system than a bank of similar size located in a state with a high concentration of financial institutions. If regionally important banks are more likely to have board representation at the Federal Reserve in their district, this scenario could explain the positive relation between Federal Reserve board membership and CPP investment.

To test this alternative hypothesis, we construct a sample of connected and unconnected banks of similar size headquartered in the same state. Specifically, for each type of connections and each connected institution, we choose an unconnected bank closest in size and domiciled in the same state. As before, we require that the difference in size between two institutions within each pair not exceed 10 percent. We then repeat the estimation of our main logit specification in each matched subsample. The estimation results, shown in Panel B of Table VI, are qualitatively similar to the findings in the main analysis. Connections to congressional representatives on finance committees and representation at the Federal Reserve via board members are positively and significantly related to the likelihood of receiving CPP capital.

Last, it is possible that bank size, even when measured within the respective geographic region, does not fully reflect the bank's role in the financial system. In particular, a bank of a given size may be more systemically important if it is interconnected with a greater number of financial institutions. The failure of such a bank, even if relatively small in the context of the entire system, may cause systemic distress as a result of the ripple effect on the stability of connected institutions.

To estimate systemic importance of individual institutions, we use Δ CoVaR – a measure of systemic risk proposed by Adrian and Brunnermeier (2009) in the Federal Reserve Bank of New York Report at the time of the credit crisis. Intuitively, this measure estimates a bank's marginal contribution to systemic risk of the financial sector. In contrast to the traditional risk measures, which focus on the individual risk of financial institutions, such as value-at-risk (VaR), expected shortfall, and the volatility of losses, Δ CoVaR captures the risk spillover across the financial network (such as the distortion in the supply of credit and capital) resulting from distress of a given financial institution.

More specifically, we define CoVaR of a particular bank as VaR of the financial sector conditional on this bank being in distress. Next, we find the difference between the computed CoVaR of an individual bank and the unconditional VaR of the financial system. This difference, Δ CoVaR, which captures the marginal addition of a particular bank to the overall systemic risk, serves as an alternative control for systemic importance in our robustness tests.

To estimate Δ CoVaR, we follow the methodology in Adrian and Brunnermeier (2009) and compute VaR and Δ CoVaR based on financial institutions' growth rates in market valued total financial assets. Formally, denote by ME_t^i the market value of an intermediary i's total equity, and by LEV_t^i the ratio of total assets to book equity. We define the normalized change in market value of total financial assets, Xt_i^t , by:

$$Xt_t^i = \frac{ME_t^i \cdot LEV_t^i - ME_{t-1}^i \cdot LEV_{t-1}^i}{ME_{t-1}^i \cdot LEV_{t-1}^i}$$

Analogously, the growth rate of market valued total assets for the financial system as a whole is given by the total market value weighted sum of the Xt_t^i across all institutions. Since the estimation of VaR and Δ CoVaR requires market data, we limit our sample to publicly traded QFIs with at least ten years of available historical data. After applying this filter, we end up with a sample of 528 QFIs, of which 282 (53.4%) were awarded CPP capital.²²

We start our estimation period in January 1986 and end in December 2008, using daily market data from CRSP and quarterly balance sheet data from COMPUSTAT. We calculate the total market value weighted sum of the Xt_t^i across all institutions in our sample and define the combined VaR of the financial system as the bottom 5 percent of the growth rate between 1986 and 2008. In a final step, for each QFI, we compute the measure of systemic risk, Δ CoVaR, as the difference between VaR of the whole system conditional on institution *i* having a growth rate equal to its own VaR and the system's unconditional VaR.

Panel C of Table VI presents the results of reestimating our base logit model – the influence of political and economic factors on CPP investment – in which we use Δ CoVaR as a proxy for banks' systemic importance. In each specification, Δ CoVaR is positively and significantly related to the likelihood of receiving CPP capital, confirming our prior evidence about the role of systemic importance in CPP decisions. The positive effect of the political variables associated with the likelihood of CPP investment – connections to the Federal Reserve and House finance committees – remains significant even in a dramatically reduced sample of

 $^{^{22}}$ In an alternative specification, we also estimate VaR and Δ CoVaR based on a broader universe of financial institutions, which also includes publicly traded broker-dealers, insurance companies, and real-estate firms (SIC codes 62, 63-64, and 65-66, respectively). Although the vast majority of these firms are ineligible for CPP participation, their inclusion in the sample enables us to estimate the interconnectedness of QFIs with firms in other sectors of financial services. Our results (unreported) are qualitatively similar.

publicly traded QFIs and after controlling for institutions' systemic risk. In column (7) of Panel C, we simultaneously control for systemic importance using both size and Δ CoVaR. The results indicate that both dimensions of systemic importance are positively related to the likelihood of CPP investment, and that even after controlling for both, connections to the Federal Reserve and House finance committees remain highly significant.

Collectively, the evidence in this section suggests that our results are unlikely to be driven by sample selection and systemic importance. Yet it is possible that other bank characteristics, possibly unobservable in our dataset, can explain the positive relation between political connections and the likelihood of receiving CPP capital. Therefore, we would like to test whether the effect of political ties extends beyond the approval for CPP funds and whether it can be identified in a subsample consisting of only successful program applicants. In the next section, we study the relation between banks' political ties and investment amounts awarded to CPP participants.

6. Extensions

This section presents extensions of our main results. We start by providing evidence on the determinants of CPP investment *amounts*. Next, we analyze how the impact of political connections varies depending on the financial strength of qualifying institutions. The section concludes with a detailed analysis of each of our political variables to identify the main drivers of their influence.

6.1 Investment Amounts

Our analysis so far has concentrated on the likelihood of investment. In this section, we analyze the *amounts* of CPP investments, which are allocated to institutions as tier-1 capital. To evaluate the relative magnitude of each investment, we follow the above classification and scale the absolute investment amount by the QFI's tier-1 capital as of September 30, 2008, the date of the
latest quarterly report of condition and income before the CPP application deadline for public and private QFIs.

To analyze the determinants of investment amounts, we estimate cross-sectional OLS regressions in which the dependent variable is the scaled CPP investment. Our sample for these tests comprises the QFIs that received CPP funds as of September 22, 2009. In these tests we did not include the 55 banks that declined the CPP investment, since we do not have reliable data on the approved investment amounts.

The independent variables are defined as in the logit regressions and include a set of four measures of political influence, a vector of Camels proxies, banks' foreclosures, and a vector of demographic variables. As in the previous models, we include regulator fixed effects and use standard errors clustered at the company level.

The results are summarized in Table VII. Columns (1)–(4) present evidence for each type of political influence estimated with bank fundamentals. Column (5) shows the joint effect of all political variables, and column (6) presents the model that includes all political, financial, and demographic variables simultaneously.

The results of estimation indicate that the drivers of investment amounts differ from those of the likelihood of CPP investment. Among the political variables, the scaled investment amount is strongly positively related to a bank's amount of political contributions and lobbying expenditures (both measured relative to asset size). This relation is significant at least at the 5 percent level and in the vast majority of specifications at the 1 percent level. The effect of these variables is also economically large. A one standard deviation increase in lobbying expenditures is associated with an additional \$10.4 million in raw CPP investment, controlling for other factors. A one standard deviation increase in political contributions to congressional candidates is associated with a \$14.6 million increase in allotted CPP funds.

The evidence so far helps explain the initially surprising finding that lobbying expenditures are not strongly associated with a higher likelihood of receiving CPP funds. It is possible that most lobbying efforts were aimed at increasing the size of CPP investments. This interpretation is supported by the fact that lobbying QFIs in our sample represent large institutions, which would likely be saved with or without lobbying. The focus of lobbying on additional funds by TARP recipients is also supported by the evidence in the financial media (Mullins and Williamson 2009). Finally, this finding is also consistent with the recent introduction of congressional legislation aimed to impose restrictions on the lobbying activity of TARP recipients.

Federal Reserve board membership and representation on congressional finance committees do not significantly explain investment amounts. This result is not entirely surprising, since bank regulators make recommendations only on the approval or denial of applications but do not determine investment amounts. Similarly, congressional finance committees, while plausibly helpful in supporting QFI applications, are unlikely to influence the amount of funds awarded.

The analysis of financial variables highlights capital adequacy as one of the fundamental drivers of investment amounts. The relation is strongly negative in all specifications. A one standard deviation decrease in capital adequacy is associated with an increase of 0.45 standard deviations in scaled CPP investment (or an additional \$52.8 million in capital), controlling for other factors. The amount of investment is also negatively related to earnings and liquidity. Collectively, these findings are consistent with an investment strategy seeking to increase the capitalization and liquidity of participating banks to an adequate level, thus providing more capital to undercapitalized, less profitable, and less liquid institutions. The allocation of greater capital amounts to weaker banks is also confirmed by the strong negative relation between

banks' financial strength index and CPP investment amounts, presented in column (7) of Table VII.

Among other variables, bank size is strongly positively related to scaled investment amounts. A one standard deviation increase in size is associated with an increase of 0.26 standard deviations in scaled investment amounts (or \$29.97 million in CPP capital), controlling for other factors. The significant positive effect of size is consistent with an investment policy aimed to provide an extra margin of safety for more systemically significant institutions.

In summary, the determinants of CPP investment amounts appear consistent with strengthening the capital position and liquidity of approved participants to an acceptable level, while providing an extra capital buffer for larger institutions. Controlling for these and other fundamental factors, banks' lobbying efforts and political donations are strongly positively related to allocated capital.

6.2 Political Influence and Financial Strength

Our main results suggest that political connections are strongly positively related to the likelihood of receiving CPP investments. In this section, we seek to understand for which types of banks these connections play the greatest role. For this analysis, we divide QFIs into terciles according to the financial strength index and investigate the influence of political variables in each tercile by interacting tercile dummies with variables of political connections. In addition to the financial strength index, which aggregates measures of condition and performance, we also control for bank size and demographics. The dependent variable in these logit regressions is an indicator equal to 1 if a bank was approved for CPP investment. As in other specifications, we include regulator-fixed effects and cluster standard errors at the company level.

The results are presented in Table VIII. The base category represents the tercile of banks with the strongest financial condition, according to the financial strength index. The interaction terms show the effects of political connections in the other two terciles. The effect of each of the four political variables is presented in the corresponding column.

For all variables of political influence, the magnitude and significance of coefficients increase virtually monotonically from the top tercile to the bottom tercile. In particular, the relation between political variables and the likelihood of CPP investment is the most pronounced for the institutions with weaker financials, which had stronger incentives to use other channels of influence.

Interpreted broadly, the evidence in this section suggests that if political ties indeed played a role in the choice of CPP investments, their effect shifted at least some capital toward less efficient, underperforming financial institutions.

6.3 Details on Congressional Committees and Federal Reserve Boards

In this section we provide additional analysis of congressional committees and board memberships at the Federal Reserve to investigate the main drivers of their influence. For congressional committees, we disentangle the effects of subcommittees to understand which groups play a more important role. For board memberships at Federal Reserve Banks, we analyze how the impact of political connections varies depending on tenure and the authority level of the board seat (main office versus a branch).

To disentangle the influence of subcommittees within the House Financial Services Committee, we assign three indicators, each denoting membership on the House Financial Services Committee, the Subcommittee on Financial Institutions, and the Subcommittee on Capital Markets, respectively. We then estimate the logit model for the likelihood of CPP investment, using indicators for subcommittees and controlling for financial and demographic variables. The results of estimation are presented in Panel A of Table IX.

The likelihood of CPP investment is positively and significantly related to representation on the Subcommittee on Capital Markets and the Subcommittee on Financial Institutions. The coefficients on both subcommittees are comparable in magnitude, suggesting roughly similar effects. This finding can be explained by the fact that both committees have oversight jurisdiction over QFIs.

The effect of membership on the House Financial Services Committee is positive but not statistically significant. This result could be attributed to the fact that the committee has six subcommittees, only two of which are directly involved in the regulation of the financial services industry. The other subcommittees are tasked with issues not directly related to banking, such as international trade and financial technology.

Next, we further analyze the effect of board seats held by QFI executives at Federal Reserve Banks. We introduce indicator variables to distinguish board seats at the main twelve banks from board seats at their branch offices and repeat the estimation of logit regressions, controlling for financial and demographic bank characteristics. The results, summarized in Panel B of Table IX, show that board seats at the main banks have somewhat larger effects on the likelihood of receiving CPP capital, compared to the board seats at branch offices. This result is consistent with the greater authority of directors at the main banks.

In column (2) of Panel B in Table IX, we consider an alternative specification of board memberships, which accounts for tenure of QFI directors. The independent variable of interest is the total number of terms served by QFI executives who were board members at Federal Reserve Banks in 2008 and 2009. As before, we include financial and demographic controls. The results indicate that the number of terms served by QFI executives at the Federal Reserve is strongly positively related to the likelihood of CPP investment, consistent with the initial hypothesis that

QFIs with more directors or with directors serving consecutive terms were more likely to be approved for CPP capital.

6.4 Details on Campaign Contributions and Lobbying Expenditures

In this section we offer further analysis of political contributions and lobbying expenditures to provide additional detail on their role in the allocation of CPP capital. For political contributions, we disentangle the effect of private donations from that of bank donations to PACs. We also study the differences between contributions to members of the House and Senate, and evaluate how the impact of donations varies with politician's tenure and seniority. For lobbying expenditures, we evaluate the power of different agencies lobbied by financial institutions and study the variation across issues involved in the lobbying process.

Panel A of Table X presents a detailed analysis of campaign contributions. Columns (1) and (2) present evidence by type of contributions (individual donations versus PAC contributions). The results indicate that the influence of donations is primarily driven by bank contributions to PACs. Private donations made by banks' employees have a weaker positive effect. This result, which holds for both House and Senate contributions, can be explained by the smaller size of individual donations and the stringent caps imposed on individual sponsorship of any particular politician (\$2,400 per election cycle). As a result, individual contributions likely yield less substantial influence for a financial institution, given their trivial magnitude and a lack of coordination at the bank level. In contrast, bank-sponsored PACs usually have a well-coordinated plan of expenditures to support particular politicians.

Columns (3) and (4) show the interaction between contributions and politician's seniority dummy (equal to one for politicians with above-median number of terms served). For both the

House and Senate specifications, the effect of contributions declines with the politician's seniority, as shown by the negative coefficient on the interaction term. This result, though somewhat surprising, may be attributed to the fact that senior politicians have greater reputation concerns and undergo higher scrutiny, which can restrain their attempts to influence the allocation of federal capital. This interpretation would be consistent with career concerns of federal officials under close monitoring (Fama 1980). Empirically, this result is also in line with the evidence that high-level politicians in the United States are often restrained by monitoring bodies and career concerns from benefiting their connected firms (for example, Fisman et al. 2006). As an alternative explanation, more senior politicians may be connected to a greater number of firms outside of the representative's district, thereby reducing the relative effect of ties to the banks in the home state.

Panel B of Table X presents additional details on lobbying expenditures, which are broken down by government agencies lobbied and issues involved. Column (1) provides evidence on total lobbying expenditures in 2008 scaled by bank size. In column (2), we use only lobbying activities in the areas of Finance, Banking, and Bankruptcy—the official classification categories most likely to include bank petitions related to TARP capital. This filter increases both the magnitude and the significance of the lobbying coefficient, as well as the R-square of the regression, supporting our interpretation of the main results.

Column (3) of Panel B in Table X uses lobbying expenditures targeted only at the Treasury and the four banking regulators—the organizations directly involved in CPP investment decisions. This filter also increases the magnitude of the lobbying coefficient, suggesting that lobbying the primary decision makers increases the positive effect on allocated capital amounts.

Collectively, the results in Panel B indicate that lobbying expenditures that are most likely to involve petitions for TARP capital have the strongest positive impact on CPP

investments. This result indicates that the relation between lobbying amounts and CPP investment is unlikely to be merely coincidental.

6.5 Effect of Presidential and Congressional Elections

The CPP was initiated several weeks before the congressional and presidential elections of 2008. The elections resulted in a change in presidential administration and a slight increase in the Democratic majority in Congress. In this section, we explore to what extent, if any, the changes in presidency and political composition of Congress had an effect on the allocation of CPP funds.

For the purposes of this analysis, we study political donations made by employees of financial institutions and their political action committees in 2008. Using political contributions data, we categorize banks into three groups: (1) banks with a strong preference toward Democrats (over 75 percent of banks' donations went to Democratic candidates); (2) banks with a strong preference toward Republicans (over 75 percent of banks' donations went to Republican candidates); and (3) banks with balanced political preference (remaining institutions). Next, we introduce dummy variables for each group and study whether the banks' political affiliation had an impact on the likelihood of receiving CPP funds or the amount received. We also study the effect of the change in presidency and in the composition of Congress that became effective in January 2009.

We are unable to reliably establish the effect of party affiliation on the allocation of CPP funds either before or after the change of administration in January 2009. For brevity, these results are not reported but are available from the authors. Several explanations for this finding are possible. First, all but the first ten transactions in CPP occurred after the results of both the presidential and congressional elections were publicly announced. Therefore, the main event of

interest—the news about a change in administration—preceded the allocation of the vast majority of CPP funds, thus making it difficult to detect its effect.

In the 2008 elections, the vast majority of seats (86 percent) were won by incumbents, and therefore, the turnover in Congress was relatively small. The changes in party affiliation of members of Congress were even smaller. Both before and after the 2008 election, Democratic candidates held the majority of seats, which increased slightly from 54 percent in the previous Congress to 59 percent in January 2009. Given the small magnitude of the effect, it is possible that our tests lack power to distill the impact of party affiliation. Finally, it is also feasible that the overall level of political connections of a financial institution dominates the affiliation with a particular party. This argument is supported by evidence that the most politically active banks, as measured by political donations, lobbying expenditures, and connections to Congress representatives, are typically connected to both Democratic and Republican parties and fall into the category of banks with balanced political preferences.

Conclusion

This paper provides empirical evidence on the determinants of capital allocation under the CPP. We consider a wide range of political, financial, and demographic indicators of qualifying financial institutions and examine how they are related to the likelihood of receiving CPP funds. Our results highlight the importance of political connections in the choice of federal investments. Ties to the Federal Reserve board members and connections with House members on finance committees are strongly positively related to the likelihood of CPP investment.

Analysis also indicates that conditional on receiving CPP funds, an institution's investment in lobbying activities and political contributions is positively related to the amount of awarded capital. Our results also suggest that political ties play a stronger role for weaker, underperforming institutions.

One possible interpretation of our findings is that our empirical design fails to control for important omitted or unobservable variables correlated with political connections but unrelated to attempts to influence investment decisions. Another interpretation is that political ties, at least in some cases, affect federal investment decisions, resulting in a deviation from the declared investment focus on healthy institutions. Under the latter interpretation, external pressures shifted at least some of the federal capital toward ailing institutions. This scenario justifies the additional requirements for accountability, disclosure, and transparency in TARP investment decisions that are advocated by the GAO and other oversight bodies of the program.

Our research complements earlier studies on the relation between political connections and firm value (Roberts 1990; Fisman 2001; Faccio 2006) by studying an important channel for this effect: preferential access to federal capital. Yet in this paper, we do not attempt to provide direct evidence on company value or the deployment of federal capital, given the short time horizon for making reliable conclusions. This is an interesting question for future research.

Appendix I

Primary Banking Regulators of Qualifying Financial Institutions

The Federal Reserve is the primary regulator for bank holding companies and state-chartered banks that are members of the Federal Reserve System.²³ Together, these institutions account for more than 90 percent of QFIs in our sample and represent over 95 percent of assets.

The FDIC has supervisory authority over state-chartered banks that are not members of the Federal Reserve System. These banks comprise the majority of institutions in the banking system, but most of them are held by bank holding companies, and their applications must be also submitted to the Federal Reserve.

The OCC regulates and supervises all national banks. In contrast to state banks, which are chartered by state banking departments and regulated by state banking laws, national banks are chartered by the OCC and regulated by federal banking laws. Further, unlike state banks, national banks are required to be investing members of the Federal Reserve System. National banks are typically larger than state banks but do not need to have nationwide operations. The decision of a bank to seek federal or state charter often depends on its preference for regulatory authorities and regulating laws, since laws in some states give state banks more authority than national banks to engage in selected nonbank activities, such as real estate investments, among others.

OTS supervises savings associations (thrifts) and savings and loan holding companies. In the past, the primary operations of savings associations were limited to accepting deposits and making residential mortgage loans. However, the differences between banks and savings associations have diminished over time, and under the current law thrifts are authorized to offer

²³ In cases where the applying institution is a bank holding company, the application is submitted both to the regulator overseeing the largest bank of the holding company and to the Federal Reserve.

largely all banking services. Savings and loan holding companies are companies that own savings associations.

Appendix II

Oversight of the Troubled Asset Relief Program

The execution of TARP, including the allocation of CPP investments, is overseen by several monitors: the Financial Stability Oversight Board (FSOB), the Congressional Oversight Panel, the Office of the Special Inspector General (SIGTARP), and the Comptroller General.

The FSOB, which consists of five members, is charged with reviewing the Treasury's actions, making recommendations to the Treasury for improvement, and preventing fraud.²⁴ The board meets monthly and provides quarterly reports to Congress. The Congressional Oversight Panel consists of five outside experts selected by majority and minority leaders in the House of Representatives and Senate.²⁵ The Congressional Oversight Panel reviews the Treasury's management of TARP and reports to Congress every month. The Comptroller General, the director of the Government Accountability Office (GAO), is responsible for monitoring program execution, providing assessment reports to Congress every two months, and auditing the program every year. SIGTARP was established by the Emergency Economic Stabilization Act (EESA). The inspector general is appointed by the president and confirmed by the Senate. As of September 30, 2009, this position was held by Neil Barofsky. The task of SIGTARP is to audit and investigate the Treasury's actions, reporting to Congress every quarter.

²⁴ FSOB members include the chairman of the Board of the Federal Reserve, secretary of the Treasury, director of the Federal Housing Finance Agency, chairman of the Securities and Exchange Commission, and secretary of the Department of Housing.

²⁵ As of September 30, 2009, members of the Congressional Oversight Panel include representatives from the banking sector, law, politics, and academia.

Appendix III

Financial Conditions of the Capital Purchase Program

Redemption of Capital

Preferred shares acquired by the Treasury can be redeemed by the QFI at the issue price. During the first three years of investment life, preferred stock may be redeemed by the QFI by issuing common or preferred stock for cash in the amount of not less than 25 percent of the issue price of preferred shares acquired by the Treasury. After the first three years of investment, preferred shares may be redeemed in whole or in part at QFIs' discretion. As of September 22, 2009, fifteen banks redeemed their preferred shares from the Treasury for a total amount of \$1.3 billion. In each case, QFIs redeemed 100 percent of preferred shares.

Participation Terms for "S" Corporations and Mutual Organizations

The terms of program participation for "S" corporations and mutual organizations are similar in spirit to those described in this section, with some modifications that accommodate the structure of these institutions.²⁶ For example, in cases of "S" corporations, the Treasury invests in senior debentures rather than in preferred stock. These debentures pay an annual coupon of 7.7 percent for the first five years and 13.8 percent thereafter, and thus have an after-tax cost of 5 percent and 9 percent, respectively, under the assumption of a 35 percent tax rate. The Treasury also receives warrants to purchase additional debentures, which are also exercised immediately.

Restrictions on Participating Institutions

During the first three years of the Treasury's ownership of preferred stock or warrants, participating banks are prohibited either from increasing dividends on their common shares or

²⁶ Additional details on the financial terms of the program for "S" corporations and mutual organizations can be obtained from the program term sheets for these institutions at http://www.financialstability.gov/roadtostability/CPPappdocs.html

from repurchasing common stock or preferred shares junior to the Treasury's investment.

Participating institutions must also impose certain restrictions on their executive compensation.²⁷

Appendix IV

Boards of Directors at Federal Reserve Banks

Each of the twelve Federal Reserve banks has a board of directors, which consists of nine members appointed for three-year staggered terms, with a maximum service tenure of two consecutive terms. Three board members in each bank are classified as Class A directors and represent the member banks in the district. Class A directors usually comprise senior executives at member institutions and are the primary focus of our analysis.

Another three directors, Class B directors, are chosen to represent the public in the respective Federal Reserve District. These directors include representatives from industry, agriculture, nonprofits, and academia. Both Class A and Class B directors are elected by member banks. In this election, all member banks of the Federal Reserve district are divided into three categories by the amount of capital: large, medium, and small. Each category of banks elects one Class A and one Class B director.

Class C directors comprise three members appointed by the Federal Reserve Board of Governors to represent the interests of labor, industry, and agriculture. Two of the Class C directors are appointed to serve as chairman and deputy chairman of the board by the Federal Reserve Board of Governors. Both Class B and Class C directors cannot be employees or directors at any bank.

²⁷ In short, participating banks are prohibited from making golden parachute payments, deducting for tax purposes compensation exceeding \$500,000 per executive, and employing compensation schemes that encourage "unnecessary and excessive risks" to the value of the institution. Banks are also required to claw back any incentive compensation paid to executives based on earnings that were subsequently restated.

Among the twelve main banks in the Federal Reserve System, eleven banks also have additional branches located in the main cities of the district. The number of branches ranges from one to four per district, depending on its size. Branches of Federal Reserve Banks have their own boards of directors.

In the majority of branches, boards consist of seven members serving three-year terms with a maximum service limit of two consecutive terms. In a typical branch board, four directors are appointed by the Reserve Bank and three are appointed by the Board of Governors, which also designates the board's chairman. Members of the branch board usually include one or two executives of member banks, with the remaining directors representing industry, agriculture, nonprofits, and academia.

Appendix V Variable Definitions

1. CPP Variables

CPP = an indicator equal to 1 if the financial institution was approved for CPP funds.

CPP amount = amount of capital received by a financial institution from the CPP, scaled by the institution's tier-1 capital.

2. Political Variables

Congress finance committee member = average of a set of indicators equal to 1 if the financial institution's congressional representatives in 2008 or 2009 were members of the Financial Services Committee, the Capital Markets subcommittee, or the Financial Institutions subcommittee.

Federal Reserve board member = an indicator equal to 1 if the financial institution had executives that are also Federal Reserve directors in 2008 or 2009.

Lobbying amount = amount of bank's lobbying expenditures from the beginning of 2008 through the first quarter of 2009.

Contribution amount = amount of contributions to congressional candidates in the 2008 election cycle, including bank contributions to PACs and individual contributions of bank employees and their spouses.

Congress representatives, state = the number of congressional representatives in the state where the financial institution is headquartered, scaled by the number of financial institutions headquartered in the state.

Congress financial services committee member = the average of two indicators equal to 1 if the company's congressional representatives in 2008 or 2009 were members of the Financial Services Committee.

Congress capital markets subcommittee member = the average of two indicators equal to 1 if the company's congressional representatives in 2008 or 2009 were members of the Congress Capital Markets subcommittee.

Congress financial institutions subcommittee member = the average of two indicators equal to 1 if the company's congressional representatives in 2008 or 2009 were members of the Financial Institutions subcommittee, respectively.

Federal Reserve board member, main office = an indicator equal to 1 if the company had executives who are also Federal Reserve directors in one of the twelve main offices in 2008 or 2009.

Federal Reserve Board member, branch = an indicator equal to 1 if the company had executives who are also Federal Reserve directors in any of the branch offices in 2008 or 2009.

Federal Reserve board member, number of terms = number of terms served by company executives who are Federal Reserve board members in 2008 or 2009.

3. Financial Variables

Capital adequacy = tier-1 risk-based capital ratio, defined as tier-1 capital divided by risk-weighted assets.

Capital adequacy refers to the amount of a bank's capital relative to the risk profile of its assets. Broadly, this criterion evaluates the extent to which a bank can absorb potential losses. Tier-1 capital comprises the more liquid subset of bank's capital, whose largest components include common stock, paid-in-surplus, retained earnings, and noncumulative perpetual preferred stock. To compute the amount of risk-adjusted assets in the denominator of the ratio, all assets are divided into risk classes (defined by bank regulators), and less risky assets are assigned smaller weights, thus contributing less to the denominator of the ratio. The intuition behind this approach is that banks holding riskier assets require a greater amount of capital to remain well capitalized.

In the robustness section, we also test an alternative specification of capital adequacy: the total risk-based capital ratio, determined as a ratio of total capital to risk-weighted assets.

Asset quality = the negative of noncurrent loans and leases, scaled by total loans and leases.

Asset quality evaluates the overall condition of a bank's portfolio and is typically evaluated by a fraction of nonperforming assets and assets in default. Noncurrent loans and leases are loans that are past due for at least ninety days or are no longer accruing interest, including nonperforming real-estate mortgages. A higher proportion of nonperforming assets indicates lower asset quality. For ease of interpretation, this ratio is included with a negative sign so that greater values of this proxy reflect higher asset quality.

In the robustness section, we also test an alternative measure: the ratio of loan and lease allowance to total loans. This ratio (also included with a negative sign) measures the adequacy of the allowance created by the bank to absorb losses on nonperforming loans.

Management quality = the negative of the number of corrective actions that were taken against bank executives by the corresponding banking regulator (FED, OTS, FDIC, and OCC) between 2006 and 2009.

In our sample, we have a total of 1,681 orders issued to 961 QFIs. The most frequent enforcement actions include prohibitions from further participation in banking activities,²⁸ orders to cease and desist,²⁹ and orders to pay civil money penalties.³⁰ To avoid subjectivity, we do not attempt to assess the severity of each corrective action but rather use the total number of enforcement orders issued to a QFI over the three-year period.

²⁸ These orders prohibit bank managers convicted of unlawful activities or unsound banking practices from involvement in any insured depository institution without prior regulatory approval.

²⁹ Cease-and-desist orders are issued when a regulator determines that a bank has engaged in "an unsafe or unsound banking practice or a violation of law." These orders require banks to take corrective actions specified by the banking regulator.

³⁰ These orders require banking organizations to pay fines for engaging in unsafe banking practices, violations of law, or incompliance with orders of banking regulators.

Earnings = return on assets (ROA), measured as the ratio of the annualized net income in the trailing quarter to average total assets. For robustness, we also measure Earnings using net interest income to earning assets.

Liquidity = cash divided by deposits. For robustness, we also measure Liquidity using the negative of the noncore dependence ratio, defined as noncore deposits divided by long-term assets.³¹

Sensitivity to market risk = the sensitivity to interest rate risk, defined as the ratio of the absolute difference (gap) between short-term assets and short-term liabilities to earning assets.

The primary focus of risk analysis by bank regulators is on interest rate risk. The gap between short-term assets and liabilities approximates the net amount of assets or liabilities that need to be repriced within one year, affecting earnings. A greater absolute value of the gap reflects a higher interest rate risk.

Foreclosures = value of foreclosed assets divided by net loans and leases.

Financial strength index = the company's average Camels ranking: *Capital adequacy, Asset quality, Management quality, Earnings, Liquidity, and Sensitivity to market risk, normalized to lie between 0 and 1.*

 $\Delta CoVaR$ = a measure of systemic risk developed by Adrian and Brunnermeier (2009). This proxy estimates a bank's marginal contribution to systemic risk of the financial sector. $\Delta CoVaR$ is the difference between CoVaR of an individual bank (i.e., value-at-risk of the financial sector conditional on this bank being in distress) and the unconditional value-of-risk of the financial system.

4. Demographic Variables

Size = the natural logarithm of total assets, defined as all assets owned by the bank holding company, including cash, loans, securities, bank premises, and other assets. This total does not include off-balance-sheet accounts.

Age = age (in years) of the oldest bank owned by the bank holding company as of September 30, 2008.

³¹ Noncore deposits represent the more volatile funds, such as brokered deposits, CDs in excess of \$100,000 and deposits by foreign institutions. These sources are most sensitive to changes in interest rates and market conditions. Since a greater reliance on noncore funds indicates lower liquidity, we take the negative of the noncore dependence ratio.

References

Adrian, Tobias and Markus K. Brunnermeier, "CoVaR," Federal Reserve Bank of New York Staff Reports, No. 348, 2009.

Aggarwal, Rajesh K., Felix Meschke, and Tracy Y. Wang, "Corporate Political Contributions: Investment or Agency?" University of Minnesota working paper, 2009.

Appelbaum, Binyamin, "Treasury to Invest in Ailing Bank," *Washington Post*, December 12, 2008.

Atkinson, Anthony B., and Joseph E. Stiglitz, *Lectures on Public Economics* (London: McGraw-Hill, 1980).

Banerjee, Abhijit V., "A Theory of Misgovernance," *Quarterly Journal of Economics*, 112 (1997), 1289–1332.

Bebchuk, Lucian A., "How to Make TARP II Work," Harvard Law and Economics Discussion Paper No. 626, 2009.

Chiu, Ming M., and Sung Wook Joh, "Loans to Distressed Firms: Political Connections, Related Lending, Business Group Affiliations, and Bank Governance," Chinese University of Hong Kong working paper, 2004.

Coates, John C., and David S. Scharfstein, "Lowering the Cost of Bank Recapitalization," *Yale Journal on Regulation*, 26 (2009), 17-29.

Cooper, Michael J., Huseyin Gulen, and Alexei V. Ovtchinnikov, "Corporate Political Contributions and Stock Returns," *Journal of Finance*, forthcoming, 2009.

Cull, Robert, and Lixin C. Xu, "Institutions, Ownership and Finance: The Determinants of Profit Reinvestment among Chinese Firms," *Journal of Financial Economics*, 77 (2005), 117–146.

Dinc, Serdar, "Politicians and Banks: Political Influences in Government-Owned Banks in Emerging Countries," *Journal of Financial Economics*, 77 (2005), 453–470.

Faccio, Mara, "Politically Connected Firms," American Economic Review, 96 (2006), 369-386.

Faccio, Mara, Ronald W. Masulis, and John J. McConnell, "Political Connections and Corporate Bailouts," *Journal of Finance*, 61 (2006), 2597–2635.

Faccio, Mara, and David C. Parsley, "Sudden Deaths: Taking Stock of Geographic Ties," *Journal of Financial and Quantitative Analysis*, 44 (2009), 683-718.

Fama, Eugene F., "Agency Problems and the Theory of the Firm," *Journal of Political Economy*, 88 (1980), 288–307.

Fisman, Raymond, "Estimating the Value of Political Connections," *American Economic Review*, 91 (2001), 1095–1102.

Fisman, David, Raymond Fisman, Julia Galef, and Rakesh Khurana, "Estimating the Value of Connections to Vice President Cheney," Columbia University working paper, 2006.

Glaeser, Edward L., and Raven Saks, "Corruption in America," *Journal of Public Economics*, 90 (2006), 1053–1072.

Goldman, Eitan, Jörg Rocholl, and Jongil So, "Do Politically Connected Boards Affect Firm Value?" *Review of Financial Studies*, 22 (2009), 2331–2360.

Hall, Robert E., and Charles I. Jones, "Why Do Some Countries Produce So Much More Output per Worker than Others?" *Quarterly Journal of Economics*, 114 (1999), 83–116.

Hart, Oliver, Andrei Shleifer, and Robert W. Vishny, "The Proper Scope of Government: Theory and an Application to Prisons," *Quarterly Journal of Economics*, 112 (1997), 1127–1162.

Harvey, Campbell R., "The Financial Crisis of 2008: What Needs to Happen after TARP," Duke University working paper, 2008.

Johnson, Simon, and Todd Mitton, "Cronyism and Capital Controls: Evidence from Malaysia," *Journal of Financial Economics*, 67 (2003), 351–382.

Keefer, Philip, and Stephen Knack, "Institutions and Economic Performance: Cross-Country Tests Using Alternative Institutional Measures," *Economics and Politics*, 7 (1995), 207–227.

Khwaja, Asim Ijaz, and Atif R. Mian, "Do Lenders Favor Politically Connected Firms? Rent Provision in an Emerging Financial Market," *Quarterly Journal of Economics*, 120 (2005), 1371–1411.

Kroszner, Randall S., and Thomas Stratmann, "Interest Group Competition and the Organization of Congress: Theory and Evidence from Financial Services' Political Action Committees," *American Economic Review*, 88 (1998), 1163–1187.

La Porta, Rafael, Florencio Lopes-de-Silanes, Andrei Shleifer, and Robert W. Vishny, "The Quality of Government," *Journal of Law, Economics and Organization*, 15 (1999), 222–279.

Mauro, Paolo, "Corruption and Growth," Quarterly Journal of Economics, 110 (1995), 681-712.

Mian, Atif R., Amir Sufi, and Francesco Trebbi, "The Political Economy of the U.S. Mortgage Default Crisis," *American Economic Review*, forthcoming, 2009.

Mullins, Brody, and Elizabeth Williamson, "Firms Keep Lobbying as They Get TARP Cash," *Wall Street Journal*, January 23, 2009.

Paletta, Damian, and David Enrich, "Political Interference Seen in Bank Bailout Decisions," *Wall Street Journal*, January 22, 2009.

Peltzman, Sam, "Toward a More General Theory of Regulation," *Journal of Law and Economics*, 19 (1976), 211–240.

Roberts, Brian E., "A Dead Senator Tells No Lies: Seniority and the Distribution of Federal Benefits," *American Journal of Political Science*, 34 (1990), 31–58.

Sapienza, Paola, "The Effects of Government Ownership on Bank Lending," *Journal of Financial Economics*, 72 (2004), 357–384.

Shleifer, Andrei, and Robert W. Vishny, "Politicians and Firms," *Quarterly Journal of Economics*, 109 (1994), 995–1025.

Stigler, George J., "The Theory of Economic Regulation," *Bell Journal of Economics*, 2 (1971), 3–21.

Stiglitz, Joseph E., "The Role of the State in Financial Markets," *Proceedings of the World Bank Annual Conference on Economic Development* (Washington, DC: International Bank for Reconstruction and Development/World Bank, 1993), 19–56.

Veronesi, Pietro, and Luigi Zingales, "Paulson's Gift," *Journal of Financial Economics*, forthcoming, 2009.

TABLE ISummary Statistics

This table reports summary statistics for the sample, which consists of all FDIC-insured, active financial institutions as of September 30, 2008. We exclude non-US banks and banks that declined to apply to CPP. All variable definitions are given in Appendix V. CPP investment is an indicator that equals 1 if the company received CPP funds or was approved and later declined the funds and 0 otherwise. Congress finance committee member is the average of a set of indicators equal to 1 if the company's Congress Representatives in 2008 or 2009 were members of the Financial Services Committee, the Capital Markets sub-committee, or the Financial Institutions sub-committee, respectively. Federal Reserve board member is an indicator that equals 1 if the company had executives that are also Federal Reserve directors in 2008 or 2009. Lobbying amount is the total amount of money invested in lobbying from the beginning of 2008 through the first quarter of 2009. Contribution amount is the total amount of contributions made to Political Action Committees (PACs), as well as individual donations initiated by banks' executives, employees, and their spouses in 2007-2008. Capital adequacy is Tier 1 risk-based capital ratio. Asset quality is the negative of noncurrent loans and leases, scaled by total loans and leases. Management quality is the negative of the number of corrective actions that were taken against bank executives by the corresponding banking regulator (FED, OTS, FDIC, and OCC) between 2006 and 2009. Earnings equal return on assets (ROA). Liquidity is measured by the cash to deposits ratio. Sensitivity to market risk is the sensitivity to interest rate risk, defined as the ratio of the absolute difference (gap) between short-term assets and short-term liabilities to earning assets. Foreclosures equal the total value of foreclosed assets scaled by net loans and leases. Age is the age of the oldest bank owned by the bank holding company in years as of 2009. Assets are all assets owned by the bank holding company including cash, loans, securities, bank premises and other assets. This total does not include off-balance-sheet accounts. Size is the natural logarithm of assets. Panel B reports CPP investment announcement returns and t-statistics (in brackets). Panel C reports pair-wise correlations.

Panel A: Summary Statistics

Variable	Mean	25th percentile	Median	75th percentile	Standard deviation
Politics					
Congress finance committee member	0.033	0.000	0.000	0.000	0.085
Federal Reserve board member	0.011	0.000	0.000	0.000	0.106
Lobbying amount, \$thousands	5,080	160	1,088	4,386	8,999
Contribution amount, \$thousands	36.228	0.708	2.000	6.050	344.937
CPP					
CPP investment indicator	0.099	0.000	0.000	0.000	0.298
CPP amount (\$thousands), recipients	301,280	4,789	11,385	33,950	1,950,750
Financials					
Capital adequacy (%)	18.416	10.763	13.423	18.628	24.211
Asset quality (%)	-1.828	-2.263	-1.022	-0.309	2.901
Management quality	-0.231	0.000	0.000	0.000	1.623
Earnings (%)	0.465	0.142	0.702	1.169	3.876
Liquidity (%)	8.247	2.587	3.872	6.623	74.465
Sensitivity to market risk (%)	13.649	2.981	8.465	18.332	33.637
Demographics					
Age	71	25	81	106	44
Assets, \$thousands	1,700,000	66,408	144,000	335,000	36,600,000

Panel B: CPP Investment Announcement Returns

Window	[-1,0]	[-1,1]	[-2,2]	
Unadjusted Raw Returns (%)	0.920 [1.877]	1.920 [3.460]	1.370 [1.980]	
Comparison Period Adjusted (%)	1.090 [2.295]	2.170 [3.973]	1.800 [2.643]	
Market Adjusted Returns, Value Weighted Index (%)	0.900 [3.622]	1.360 [3.650]	0.940 [2.155]	
Industry Adjusted Returns (%)	1.130 [4.519]	1.810 [5.207]	1.610 [3.523]	
Fama-French Three-Factor Model (%)	1.280 [3.516]	2.510 [4.837]	2.990 [4.218]	

Panel C: Correlation Matrix

Variable	Congress finance committee member	Federal Reserve board member	Lobbying amount	Contribution amount	Congress Reps, state	Capital adequacy	Asset quality	Managers' quality	Earnings	Liquidity	Sensitivity to market risk	Age	Size
Congress finance committee member	1.000												
Federal Reserve board member	0.024	1.000											
Lobbying amount	-0.010	0.003	1.000										
Contribution amount	0.037	0.142	0.232	1.000									
Congress Reps, state	0.058	0.025	0.029	0.064	1.000								
Capital adequacy	0.009	-0.030	0.050	0.004	0.052	1.000							
Asset quality	-0.023	-0.002	0.007	0.001	-0.032	0.096	1.000						
Management quality	-0.034	-0.178	-0.008	-0.618	-0.048	0.038	0.095	1.000					
Earnings	-0.026	0.004	0.001	0.000	-0.116	0.183	0.116	0.026	1.000				
Liquidity	0.007	0.023	0.009	0.116	0.012	0.237	0.010	-0.159	0.099	1.000			
Sensitivity to market risk	0.013	-0.001	-0.010	0.008	0.045	0.122	0.039	0.008	0.022	0.144	1.000		
Age	-0.060	0.022	-0.055	0.013	-0.283	-0.077	0.091	-0.027	0.101	-0.028	0.021	1.000	
Size	0.063	0.197	0.088	0.229	0.237	-0.260	-0.085	-0.262	-0.008	-0.028	-0.024	-0.011	1.000

TABLE II

Geography and CPP Funds Allocation

This table presents mean estimates of the likelihood of receiving CPP funds and the amounts of CPP funds across quintiles of state-level Congress representation. Congress representation is measured by Congress Representatives, state, defined as the number of Congress Representatives for the company's state, scaled by the number of banks headquartered in the state (Panel A) or by the average size of the banks headquartered in the state (Panel B). The sample consists of all FDIC-insured, active financial institutions as of September 30, 2008. We exclude non-US banks and banks that declined to apply to CPP.

Panel A: Number of Congress Representatives / number of banks headquartered in the state

Congress Representatives, state	Likelihood	Amount
Low quintile	0.039	1,481
2nd quintile	0.071	9,223
3rd quintile	0.093	13,572
4th quintile	0.106	15,756
High quintile	0.184	91,497
Difference (High-Low)	0.145	90,017
T-Statistic for Difference (High-Low)	12.452	2.601

Panel B: Number of Congress Representatives / average size of the banks headquartered in the state

Congress Representatives, state	Likelihood	Amount
Low quintile	0.072	6,234
2nd quintile	0.087	5,902
3rd quintile	0.108	12,755
4th quintile	0.108	28,674
High quintile	0.121	85,447
Difference (High-Low)	0.049	79,213
T-Statistic for Difference (High-Low)	4.368	2.322

TABLE III

Non Parametric Evidence: The Likelihood of Receiving CPP Funds

This table presents difference-in-means estimates of the likelihood of receiving CPP funds. The sample consists of all FDIC-insured, active financial institutions as of September 30, 2008. We exclude non-US banks and banks that declined to apply to CPP. The financial strength index is the company's average ranking on Capital adequacy, Asset quality, Management quality, Earnings, Liquidity, and Sensitivity to market risk (Camels), normalized to lie between 0 and 1. See Appendix V for all variable definitions.

	Low	High	High Minus Low	t-statistic
Politics				
Congress finance committee member	0.098	0.206	0.108	2.557
Federal Reserve board member	0.094	0.470	0.376	11.518
Lobbying amount	0.096	0.366	0.270	7.633
Contribution amount	0.074	0.204	0.123	14.698
Financials				
Capital adequacy	0.169	0.028	-0.141	20.716
Asset quality	0.114	0.083	-0.031	4.460
Management quality	0.127	0.094	-0.033	3.199
Earnings	0.128	0.069	-0.059	8.514
Liquidity	0.126	0.071	-0.054	7.795
Sensitivity to market risk	0.104	0.093	-0.011	1.619
Foreclosures	0.079	0.118	0.039	5.561
Financial strength index	0.161	0.036	-0.124	18.166
Demographics				
Age	0.126	0.071	-0.054	7.810
Size	0.031	0.166	0.135	19.767

TABLE IV The Likelihood of CPP Investment

banks that declined to apply to CPP. See Appendix V for all variable definitions. All regressions include regulator fixed effects. The t-statistics (in brackets) are based on standard errors that are heteroskedasticity consistent and clustered at the company level. Significance levels are indicated: * = 10%, ** = 5%, *** = 1%. institution was approved for CPP funds. The sample consists of all FDIC-insured, active financial institutions as of September 30, 2008. We exclude non-US banks and This table presents estimates from logit regressions explaining the likelihood of CPP investment. The dependent variable is an indicator equal to 1 if the financial

Model Number	(1)	(2)	(3)	(4)	(5)	(9)	(2)
Congress finance committee member	1.203*** [2.627]				1.713*** [4.172]	1.172** [2.553]	0.855** [1.998]
Federal Reserve board member		0.722** [2.534]			1.876*** [7.885]	0.716^{**} [2.504]	0.675** [2.366]
Lobbying amount			1.488 [1.345]		1.122 [0.926]	2.373 [1.474]	1.711 [1.491]
Contribution amount				0.243* [1.857]	0.854*** [3.457]	0.249* [1.904]	0.242^{*} $[1.953]$
Capital adequacy	-0.125*** [9.454]	-0.124*** [9.426]	-0.122*** [9.263]	-0.125*** [9.420]		-0.124*** [9.345]	
Asset quality	0.201^{***} [6.941]	0.201^{***} [6.927]	0.201^{***} [6.938]	0.199^{***} [6.870]		0.200^{***} [6.875]	
Management quality	0.072*** [2.975]	0.076*** [3.255]	0.073*** [3.087]	0.084^{***} [2.815]		0.100^{**} [3.208]	
Earnings	-0.149*** [4.717]	-0.153*** [4.852]	-0.152*** [4.804]	-0.154*** [4.845]		-0.151*** [4.750]	
Liquidity	0.002 [0.948]	0.002 [0.952]	0.002 [1.050]	0.002 [0.814]		0.001 [0.409]	
Sensitivity to market risk	-0.007* [1.837]	-0.006* [1.651]	-0.007* [1.700]	-0.007* [1.702]		-0.007* [1.749]	
Foreclosures	-0.003*** [3.531]	-0.003*** [3.525]	-0.003*** [3.573]	-0.003*** [3.559]		-0.003*** [3.521]	-0.004*** [6.181]
Age	-0.007*** [7.078]	-0.008*** [7.291]	-0.008*** [7.353]	-0.007*** [7.192]		-0.008*** [7.168]	-0.009*** [8.604]
Size	0.739^{***} [19.502]	0.729*** [19.093]	0.751^{***} [19.696]	0.740^{***} [19.379]		0.727*** [18.695]	0.719^{***} [20.016]
Financial strength index							-2.794*** [7.093]
Regulating agency F.E?	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Pseudo R ²	0.244	0.244	0.243	0.243	0.062	0.246	0.210
N_obs	7,255	7,255	7,255	7,255	7,255	7,255	7,255

TABLE V Robustness to Sample Selection

approved for CPP funds. The sample consists of all FDIC-insured, active financial institutions as of September 30, 2008. We exclude non-US banks and banks that declined to apply to CPP. Previous variable definitions correspond to all columns but (1) and (2). In column (1), Capital adequacy is total risk-based capital ratio, Asset quality is the opposite of net losses to average total loans and leases, Earnings is noninterest income to earning assets, and liquidity is the opposite of the non-core dependence ratio, defined as non-core This table presents estimates from logit regressions explaining the likelihood of CPP investment. The dependent variable is an indicator equal to 1 if the financial institution was Earnings is net interest income to earning assets, and liquidity is core deposits to asset growth. See Appendix V for all variable definitions. All regressions include regulator fixed effects. The t-statistics (in brackets) are based on standard errors that are heteroskedasticity consistent and clustered at the company level. Significance levels are indicated: * = deposits divided by long-term assets. In column (2), Capital adequacy is the tier 1 leverage capital ratio, Asset quality is the opposite of loan and lease allowance to total loans, 10%, ** = 5%, *** = 1%.

Description	Different Controls	Different Controls	Exclude top 25% capital adequacy	Exclude top 25% quality index	Exclude top 1% size	Exclude NY
Model Number	(1)	(2)	(3)	(4)	(5)	(9)
Congress finance committee member	1.140**	1.064**	1.193**	1.228**	0.969**	1.429***
	[2.508]	[2.350]	[2.517]	[2.545]	[2.053]	[2.976]
Federal Reserve board member	0.721**	0.672**	0.716**	0.720**	0.594*	0.704**
	[2.561]	[2.374]	[2.490]	[2.452]	[1.921]	[2.444]
Lobbying amount	2.566	2.884*	2.296	1.691	0.563	2.108
	[1.639]	[1.930]	[1.329]	[1.250]	[0.475]	[1.262]
Contribution amount	0.337	0.166	0.166	0.417	1.989	0.033
	[1.499]	[1.053]	[0.651]	[1.287]	[1.366]	[0.101]
Capital adequacy	-0.114^{***} [8.698]	-0.066*** [4.412]	-0.148*** [7.093]	-0.110*** [7.121]	-0.129*** [9.343]	-0.128*** [9.205]
Asset quality	0.000 [0.251]	0.266^{***} [3.478]	0.193^{***} [6.380]	0.205*** [6.883]	0.183^{***} [6.114]	0.220*** [7.266]
Management quality	0.100^{**}	0.092^{***}	0.092***	0.127***	0.399^{***}	0.093**
	[3.131]	[3.190]	[2.749]	[3.396]	[4.178]	[2.342]
Earnings	-0.051	0.047	-0.081**	-0.107***	-0.176***	-0.174***
	[1.359]	[1.152]	[2.191]	[3.337]	[5.285]	[5.264]
Liquidity	0.000	0.000	0.001	0.001	-0.029**	0.003*
	[1.407]	[1.217]	[0.549]	[0.577]	[2.508]	[1.812]
Sensitivity to market risk	-0.006	-0.013***	-0.004	-0.011***	-0.007*	-0.008*
	[1.421]	[3.317]	[0.886]	[2.699]	[1.784]	[1.929]
Foreclosures	-0.004*** [5.834]	-0.003*** [4.342]	-0.002^{**} [3.169]	-0.003*** [3.659]	-0.003*** [3.674]	-0.003*** [3.613]
Age	-0.008***	-0.010^{**}	-0.007***	-0.007***	-0.008***	-0.008***
	[7.542]	[9.175]	[6.408]	[6.338]	[7.476]	[7.327]
Size	0.709^{***} [18.992]	0.807^{***} [21.511]	0.722^{***} [18.051]	0.718^{***} [17.513]	0.798^{**} [17.825]	0.747^{***} [18.667]
Regulating agency F.E?	Yes	Yes	Yes	Yes	Yes	Yes
Pseudo R ²	0.231	0.215	0.203	0.224	0.237	0.247
N_obs	7,255	7,255	5,441	5,441	7,182	7,082

	Importance
FABLE VI	o Svstemic
	Robustness t

size. In Panel B, banks are matched by size to banks headquartered in their state. In both panels, matched pairs in which the size difference between two banks is calculation of the $\Delta CoVar$ measure is based on market values. Connected is an indicator equal to 1 if the company is politically connected according to any of the institution was approved for CPP funds. The baseline sample consists of all FDIC-insured, active financial institutions as of September 30, 2008. We exclude according to the political connections measure in the title of the column and their best-match unconnected counterpart. In Panel A, banks are matched by their defined as the value at risk of the financial system conditional on institutions being under distress. The sample is limited to publicly traded institutions, as the political connections measures. See Appendix V for all other variable definitions. All regressions include regulator fixed effects. The t-statistics (in brackets) are This table presents estimates from logit regressions explaining the likelihood of CPP investment. The dependent variable is an indicator equal to 1 if the financial non-US banks and banks that declined to apply to CPP. In each column of Panels A and B, we consider subsample of all banks that are politically connected greater than 10% are excluded from the analysis. In Panel C, systemic importance is measured by $\Delta CoVaR$, suggested by Adrian and Brunnermeier (2009), and based on standard errors that are heteroskedasticity consistent and clustered at the company level. Significance levels are indicated as follows: * = 10%, ** = 5%, *** = 1%.

Panel A: Size-Matched Sample					
Sample Criterion (Measure of	Congress finance	Federal Reserve	Lobbying	Contribution	Connected
Political Connections)	committee member	board member	amount	amount	
Political connections	1.780*** [3.102]	0.222** [2.389]	0.248 [0.196]	0.012 [0.020]	2.157*** [2.655]
Regulating agency F.E?	Yes	Yes	Yes	Yes	Yes
Controls?	Yes	Yes	Yes	Yes	Yes
Pseudo R ²	0.202	0.417	0.302	0.144	0.157
N_obs	2,358	154	82	2,596	4,450
T-statistic for the difference in mean size across matched banks	0.014	0.019	0.012	0.041	0.018
Panel B: State-Size-Matched Sam	ple				
Sample Criterion (Measure of Political Connections)	Congress finance committee member	Federal Reserve board member	Lobbying amount	Contribution amount	Connected
Political connections	2.192*** [3.471]	0.716** [2.504]	1.892 [0.628]	0.088 [0.083]	2.559*** [2.598]
Regulating agency F.E?	Yes	Yes	Yes	Yes	Yes
Controls?	Yes	Yes	Yes	Yes	Yes
Pseudo R ²	0.172	0.503	0.566	0.166	0.170
N_obs	2,144	66	52	2,128	3,786
T-statistic for the difference in mean size across matched banks	0.008	0.030	0.030	0.067	0.076

Panel C: ∆CoVaR							
Model Number	(1)	(2)	(3)	(4)	(5)	(9)	(7)
ACoVaR	3.103^{***}	2.963***	3.009***	3.176^{***}	3.147 * * *	2.379**	2.436**
	[3.814]	[3.640]	[3./16]	[3.8/3]	[<i>3.</i> /8/] 1.000**	[2.037] 2.036**	[2.060] 2.055**
Congress finance committee member	1.942^{**} [2.130]				1.898** [2.062]	2.948** [2.545]	2.955** [2.506]
Federal Reserve board member		1.694*** [3.094]			1.619*** [2.932]	1.290** [2.175]	1.085** [2.173]
Lobbying amount			0.101 0.0831		2.082 [0 800]	4.585 11.7651	6.760 11.7381
			[con.n]	0.255	0.305 [0.699]	[1.202] 0.813***	0.830**
Contribution amount				[1.623]	[1.623]	[2.590]	[2.130]
Camital adenuacy						-0.408***	-0.381***
Capital auxiliary						[7.089]	[6.548]
Asset quality						0.222*** [2.883]	0.235*** [3 070]
						[00]	0.156***
Management quality						[2.580]	[3.150]
Horninge						0.347***	0.338^{***}
Lanuago						[3.677]	[3.625]
Liquidity						-0.001	0.000
Gininhir						[0.360]	[0.008]
Sensitivity to market risk						-0.026***	-0.022**
						[2.599]	[2.190]
Foreclosures						-0.003 [1.297]	-0.002 [1.074]
						0.001	-0.001
Age						[0.301]	[0.234]
Size							0.326^{***} [3.133]
Regulating agency F.E?	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Pseudo R ²	0.061	0.073	0.055	0.063	0.086	0.288	0.303
N_obs	528	528	528	528	528	528	528

TABLE VII CPP Investment Amount

This table presents estimates from OLS regressions where the dependent variable is the amount of capital received by a financial institution from the CPP, scaled by the institution's tier-1 capital. The sample consists of all financial institutions that received capital from CPP. See Appendix V for all variable definitions. All regressions include regulator fixed effects. The t-statistics (in brackets) are based on standard errors that are heteroskedasticity consistent and clustered at the company level. Significance levels are indicated: * = 10%, ** = 5%, *** = 1%.

Model Minuber	(1)	6	0	(4)	(5)	(2)	
	-0.078	(7)	(C)	(+)	-0.017	-0.007	-0.046
Congress finance committee member	[0.199]				[0.115]	[0.013]	[0.317]
Federal Reserve board member		0.020 [0.284]			0.100 [1.416]	0.042 [0.631]	0.018 [0.253]
Lobby amount			2.292*** [7.810]		1.728*** [4.945]	1.718*** [4.967]	1.754*** [5.049]
Contribution amount				0.117^{***} [6.729]	0.032** [2.023]	0.064*** [3.144]	0.014^{***} [2.831]
Capital adequacy	-4.867*** [9.109]	-4.861*** [9.097]	-5.476*** [10.600]	-5.316*** [10.208]		-5.560*** [10.801]	
Asset quality	2.424** [2.161]	2.441** [2.175]	1.924* [1.792]	1.684 [1.546]		1.673 [1.560]	
Management quality	-0.644 [1.140]	-0.627 [1.104]	-0.7 <i>5</i> 7 [1.401]	1.218^{**} $[1.990]$		0.316 [0.498]	
Earnings	-2.365* [1.952]	-2.330* [1.932]	-2.768** [2.400]	-2.174* [1.866]		-2.550** [2.208]	
Liquidity	-0.098*** [2.611]	-0.098*** [2.615]	-0.098*** [2.740]	-0.087** [2.417]		-0.092*** [2.589]	
Sensitivity to market risk	0.067 [0.470]	0.066 [0.464]	-0.052 [0.379]	0.042 [0.303]		-0.034 [0.247]	
Foreclosures	-0.016 [0.576]	-0.016 [0.559]	-0.014 [0.519]	-0.019 [0.698]		-0.015 [0.585]	-0.015 [0.521]
Age	-0.075** [2.180]	-0.076** [2.192]	-0.044 [1.334]	-0.064* [1.918]		-0.047 [1.421]	-0.035 [0.996]
Size	2.769** [2.455]	2.699** [2.348]	1.205 [1.099]	1.042 [0.931]		0.525 [0.466]	3.635*** [3.119]
Financial strength index							-0.398** [2.357]
Regulating agency F.E?	Yes	Yes	Yes	Yes	Yes	Yes	Yes
\mathbb{R}^2	0.183	0.183	0.253	0.236	0.077	0.265	0.108
N_obs	660	660	660	660	660	660	660

TABLE VIII

Political Connections and Financial Strength

This table presents estimates from logit regressions explaining the likelihood of CPP investment. The dependent variable is an indicator equal to 1 if the financial institution was approved for CPP funds. The sample consists of all FDIC-insured, active financial institutions as of September 30, 2008. We exclude non-US banks and banks that declined to apply to CPP. The financial strength index is the company's average ranking on Capital adequacy, Asset quality, Management quality, Earnings, Liquidity, and Sensitivity to market risk (Camels), normalized to lie between 0 and 1. See Appendix V for all variable definitions. All regressions include regulator fixed effects. The t-statistics (in brackets) are based on standard errors that are heteroskedasticity consistent and clustered at the company level. Significance levels are indicated: * = 10%, ** = 5%, *** = 1%.

Explanatory Variable (var)	Congress finance committee member	Federal Reserve board member	Lobbying amount	Contribution amount
Model Number	(1)	(2)	(3)	(4)
var	-0.699	-0.386	-15.579	-3.746
	[0.548]	[0.316]	[1.030]	[1.089]
var x medium strength index	1.437	1.025*	13.935	3.681*
	[0.972]	[1.791]	[0.918]	[1.657]
var x low strength index	2.087**	1.269**	17.019*	3.677*
	[1.995]	[1.994]	[1.718]	[1.657]
Financial strength index	-1.653***	-1.836***	-1.732***	-1.661***
	[4.286]	[4.984]	[4.717]	[4.474]
Age	-0.008***	-0.008***	-0.008***	-0.008***
	[7.910]	[8.085]	[8.211]	[8.247]
Size	0.749***	0.736***	0.765***	0.775***
	[21.736]	[21.066]	[21.903]	[21.653]
Regulating agency F.E?	Yes	Yes	Yes	Yes
Pseudo R ²	0.195	0.195	0.196	0.196
N_obs	7,255	7,255	7,255	7,255

TABLE IX Extensions: The Likelihood of Receiving CPP Funds

This table presents estimates from logit regressions explaining the likelihood of CPP investment. The dependent variable is an indicator equal to 1 if the financial institution was approved for CPP funds. The sample consists of all FDIC-insured, active financial institutions as of September 30, 2008. We exclude non-US banks and banks that declined to apply to CPP. Panel A considers Congressional committees. Congress financial services committee member is the average of 2 indicators equal to 1 if the company's Congress representatives in 2008 or 2009 were members of the Financial Services Committee. Congress financial institutions sub-committee member and Congress capital markets sub-committee member are defined analogously with respect to the Capital Markets sub-committee and the Financial Institutions sub-committee, respectively. Panel B considers Federal Reserve board members. Federal Reserve board member, main office is an indicator that equals 1 if the company had executives that are also Federal Reserve directors in one of the 12 main offices in 2008 or 2009. Federal Reserve board member, branch is defined analogously with respect to directors in the different branch offices. Federal Reserve board member, number of terms is the total number of terms served by company executives that are Federal Reserve board members in 2008 or 2009. The list of control variables is similar to that of previous tables. See Appendix V for all other variable definitions. All regressions include regulator fixed effects. The t-statistics (in brackets) are based on standard errors that are heteroskedasticity consistent and clustered at the company level. Significance levels are indicated: * = 10%, ** = 5%, *** = 1%.

Panel A: Congressional	committees
-------------------------------	------------

Model Number	(1)	(2)	(3)
Congress financial services committee member	0.188 [0.506]		
Congress financial institutions sub-committee member		0.677** [2.404]	
Congress capital markets sub-committee member			0.540*** [3.295]
Control variables?	Yes	Yes	Yes
Regulating agency F.E?	Yes	Yes	Yes
R^2	0.243	0.246	0.246
N_obs	7,255	7,255	7,255

Panel B: Federal Reserve board members

Model Number	(1)	(2)
Federal Reserve board member, main office	0.768*** [2.252]	
Federal Reserve board member, branch	0.695** [2.211]	
Federal Reserve board member, number of terms		0.483*** [3.459]
Control variables?	Yes	Yes
Regulating agency F.E?	Yes	Yes
R^2	0.244	0.245
N_obs	7,255	7,255

TABLE XExtensions: CPP Amounts

This table presents estimates from OLS regressions where the dependent variable is the amount of capital received by a financial institution from the CPP, scaled by the institution's tier-1 capital. The sample consists of all financial institutions that received capital from CPP. Panel A considers contribution amounts. PAC amounts are amounts contributed through Political Action Committees. Private amounts are amounts contributed by individuals working for the company. Senior is an indicator variable equal to 1 if the house member or senator has longer-than-median tenure. Panel B considers lobbying amounts. Lobbying amount, issues only considers lobbying for the following issues: Finance, Banking, and Bankruptcy. Lobbying amount, agency only considers amounts spent lobbying the Treasury and the banking regulators (FED, OTS, FDIC, and OCC). The list of control variables is similar to that of previous tables. See Appendix V for all other variable definitions. All regressions include regulator fixed effects. The t-statistics (in brackets) are based on standard errors that are heteroskedasticity consistent and clustered at the company level. Significance levels are indicated: * = 10%, ** = 5%, *** = 1%.

	Congress	Senate		Congress	Senate
Model Number	(1)	(2)		(3)	(4)
PAC contribution amount	0.256** [2.603]	0.519*** [3.849]			
Private contribution amount	0.441* [1.775]	0.140* [1.759]			
Contribution amount				3.501*** [6.975]	1.945*** [6.475]
Contribution amount x senior				-4.205*** [6.369]	-2.544*** [5.891]
Control variables?	Yes	Yes		Yes	Yes
Regulating agency F.E?	Yes	Yes		Yes	Yes
R^2	0.231	0.238		0.276	0.277
N_obs	660	660		660	660
Panel B: Lobbying Amounts					
Model Number		(1)	(2)		(3)
Lobbying amount, total		2.292*** [7.810]			
Lobbying amount, issues			3.943 [9.553	*** 3]	
Lobbying amount, agencies					3.480*** [6.439]
Control variables?		Yes	Yes		Yes
Regulating agency F.E?		Yes	Yes		Yes
R^2		0.253	0.284		0.232

660

660

660

Panel A: Contributions

N_obs

FIGURE I

State-by-State CPP Investments This Figure presents the geographic distribution of CPP investments in banks across U.S. states. The data on the recipients of CPP funds are collected from the Office of Financial Stability, which provides the amount of funds invested, the date of investment, and the consideration received by the Treasury in exchange for capital.

Panel A: Number of Banks with CPP Investments



Panel B: Number of Banks with CPP Investments/Total Number of Banks



Panel C: Total Amount of CPP Investments



Panel D: Total Amount of CPP Investments/Average banks Size

